THE ROLE OF FAMILY FUNCTIONING IN THE DECISION-MAKING STYLES OF ADOLESCENTS IN THE OVERBERG AREA

Suzette René Lenders

Student Number: 3374207

A full thesis for the fulfillment of the requirements for the Degree of

MASTERS IN CHILD AND FAMILY STUDIES DEPARTMENT OF SOCIAL WORK FACULTY OF COMMUNITY AND HEALTH SCIENCES UNIVERSITY OF THE WESTERN CAPE

Supervisor: Professor N Roman

May 2015

ABSTRACT

Adolescence has been regarded as a particularly important stage for the continued development of young adults as they start to make their own life's decisions and engage in a variety of activities that will influence the rest of their lives. One of the important goals of adolescence is the development of decision-making skills as independence increases, requiring more independent decisions. When an adolescent makes decisions, it is important that s/he has the skills to make confident ones.

Various studies identify factors that could affect adolescent decision-making, such as the family environment and family-member relationships, but information pertaining to this subject is still relatively limited. This study examines the role of family functioning in the decision-making styles of adolescents. The theoretical framework used for this study is based on the Family Systems Theory, of which the McMaster Model of Family Functioning (MMFF) is a component. The MMFF is one variation that underlies the family system model, be it nuclear or extended families discussed within chapter 2.

A quantitative methodological approach was employed in this study with a cross-sectional correlational research design. The sample consisted of 457 Grade 9 learners from schools in the Overberg area. The data was collected using a self-reported questionnaire that included the Demographic Information, the Family Assessment Device and the Melbourne Decision Making Questionnaire, as part of the quantitative methodology. The data was then analysed using the Statistical Package for Social Sciences V22 (SPSS). The results were provided using descriptive and inferential statistics.

Participation in this study was voluntary after being well informed, while confidentiality and anonymity was maintained throughout the study. The main results for family functioning indicated the assumption that the participants might have ineffective functioning on all the family functioning variables. The main results for decision-making styles showed an assumption that the Vigilant decision-making style was the highest and the Buck-passing decision-making style, the lowest.



KEYWORDS

Adolescence

Decision-making

Decision-making style

Family functioning

Problem solving

Communication

Roles

Affective responsiveness

Affective involvement



ABBREVATIONS

DMQ - Decision-Making Questionnaire

FAD - Family Assessment Device

M - Means

MMFF - McMaster Model of Family Functioning

SD - Standard Deviations

SPSS - Statistical Package for Social Sciences

GFF - General Family Functioning



DECLARATION

I declare that, The role of family functioning in the decision-making styles of adolescents in

the Overberg area, is my original work, that it has not been submitted for any degree or

examination at any other University and that all the sources I have used or quoted have been

indicated and acknowledged by complete references.

Full names: Suzette René Lenders

Date:....

WESTERN CAPE

Signed:

vi

ACKNOWLEDGEMENTS

This thesis would not have been possible without the grace of God, who, through his mercy, has granted me the resilience to come this far in my academic achievement.

To my family and friends, I am eternally grateful for all the support and motivation. This has been a great experience and a learning curve. Consequently, I feel truly enriched, to which nothing else could ever compare.

To my husband, my parents and parents-in-law, who were always available to assist and support when the going got tough. Thank you for always being there for me.

To my supervisor, Professor N. Roman, thank you for your continued encouragement and constructive criticism. You have helped to keep my sanity intact when it seemed as though I was going to fall apart. Mostly, you have enriched me with so much knowledge.

To my fellow project member, Eugene Davids, thank you for your guidance, support and for always being there to help when needed.

TABLE OF CONTENTS

CHAPTER 1

INTRODUCTION

1.1. Background and Rationale		1
1.2. Theoretical Framework		4
1.3. Problem Statement		5
1.4. Research Questions		6
1.4.1. Aim of the study		6
1.4.2. Objectives of the	Study	6
	UNIVERSITY of the	
1.5. Hypothesis of the study	WESTERN CAPE	7
1.6. Methodology		7
1.7. Significance of the study a	nd conclusion	8
1.8. Definitions and description	ns of key concepts	8
1.9. Summary of the chapters		11

THEORETICAL FRAMEWORK

2.1. Introduction	13
2.2. Defining and describing the family	13
2.3. South African families	16
2.4. Family systems theory	18
2.5. The McMaster model of family functioning	23
2.5.1. Dimensions of the McMaster Model of Family Functioning	26
2.6. Family as a caring and support network	28
2.7. South African Family Policy WESTERN CAPE	30
2.8. Conclusion	35

LITERATURE REVIEW

3.1. Introduction	36
3.2. Adolescence and their development	36
3.2.1. Adolescence and autonomy	41
3.3. Decision making	43
3.3.1. The decision making process	44
3.3.2. Decision making styles	46
3.3.3. Decision making and adolescents	48
3.3.4. Relational aspects of decision making	50
3.4. The influence of family functioning	51
3.5. Family functioning and adolescence	52
3.6. Conclusion	55

RESEARCH FINDINGS

4.1. Introduction	56
4.2. Objectives of the study	56
4.3. Research methodology	57
4.4. Research design	57
4.5. Study population	58
4.6. Study Sample	59
4.7. Instruments	60
4.7.1. The Family Assessment Device	the PE
4.7.2. Melbourne Decision Making Questionnair	re 61
4.8. Data collection for the pilot study	62
4.9. Results	63
4.10. Data collection for the main study	64
4.11. Data analysis	64
4.12. Reliability and validity	
4.13. Ethics statement	66
4.14. Conclusion	67

RESULTS

5.1. Introduction		68
5.2. Overview of the analysis		68
5.3. Inter-correlations		69
5.4. Demographic Profile		69
5.5. Family Functioning		71
5.6. Decision-making styles		80
5.7. Relationships between the	variables	84
5.8. Conclusion	UNIVERSITY of the	88

DISCUSSION, CONCLUSION AND RECOMMENDATIONS

6.1. Introduction		89
6.2. Family Functioning and add	plescence	89
6.3. Decision making styles of a	adolescence	98
6.4. Relating family functioning	to decision making styles during adolescence	102
6.5. The implications of these re	esults for the adolescent	104
6.6. Theoretical implications		107
6.7. Limitations of the study		111
6.8. Recommendations	UNIVERSITY of the WESTERN CAPE	111
6.9. Conclusion		112

REFERENCES

A-Z 114-127



APPENDICES

Appendix I: Questionnaire	96
Appendix II: Parent's Consent Form	106
Appendix III: Assent Form for Participant	107
Appendix IV: Information Sheet / Letter	108
Annendix V: Editorial certificate	111



INTRODUCTION

1.1. Background and Rationale

Adolescence is viewed as a period in life when most of an individual's biological, cognitive, psychological and social characteristics are changing towards becoming an adult (Ogwo, 2013). It is a unique developmental period of transition from childhood to adulthood (McKinney, Donelly & Renk, 2008). There is the desire to become less dependent on parents, while pursuing opportunities to be more active in decision-making (Louw, Van Ede & Louw, 1998). This period of development is also regarded as stressful and is commonly referred to as a period of storm, stress and the need for experimentation (Roman & Frantz, 2013), characterized by hormonal, physical, emotional and cognitive changes (Alloy, Zhu & Abramson, 2003). During this phase individuals have to learn to be socially responsible for themselves and for their actions (Ogwo, 2013).

Zarrett & Eccles (2006) considers this phase particularly important for the continued development of adolescents as they start to encounter a variety of choices and subsequently engage in decision-making activities that will influence the rest of their lives. The decisions that have to be made are increasingly important as it could permanently affect the adolescents' path into adulthood (Smith, Xiao & Bechara, 2011). This ability to acquire a sense of decision-making competence, according to Seiffge-Krenke (2002), may have far reaching consequences to many of life's contexts for adolescents. These choices and challenges include decisions about education or vocational training, entry into and transitions

within the labour market, moving out of the family home and, sometimes, marriage and parenthood (McKinney et al., 2008).

An important goal of adolescence is the development of decision-making skills, especially as an adolescent's increasing need for independence and autonomy involves making more decisions on their own (Miller & Byrnes, 2001). Adolescent decisions are based on situations and ideas of what is possible and imagined. They are generally made under conditions of emotional arousal, whether positive or negative (Smith et al., 2011). It is, therefore, important to chart the development of decision-making throughout adolescence (Miller & Byrnes, 2001), and vital to realize that adolescents with an effective decision-making style may experience more satisfaction in their lives (Bicanli, 2000, as cited in Deniz, 2006).

According to Steinberg (2005), the adolescent's decision-making capability improves with maturity and develops more competence. During this stage of formal operations, adolescents develop the ability to think abstractly and theoretically, and a complete transition through this stage results in the ability to: draw conclusions from available information and predict possible outcomes; reason theoretically; and understand more abstract concepts, such as love and values (Fantasia, 2008).

Decision-making, according to Miller & Byrnes (2001), can be defined as the process of choosing between different alternatives, while in the midst of pursuing a goal. It does not happen in isolation, in fact, there are a number of elements in the individual's immediate environment that play a role in decision-making. Factors that affect decision-making could include the family environment and the relationships that individuals have with family members (Armesh, 2013).

The family, however, is considered to be the primary setting in which children begin to acquire their beliefs, attitudes, values and behaviour (Ogwo, 2013). The family is the support system that adolescents turn to when they are faced with fears and doubts about themselves and their future (Caprara, Pastorelli, Regalia, Scabini & Bandura, 2005, as cited in Koen, Van Eden & Venter, 2011). The role of the family, therefore, seems crucial in adolescent development (Koen et al., 2011) as the family is perceived as a working unit that is greater than the sum of its parts; each member affects the family as a whole, while the family members also affect each individual member (Buswell, Zarbriskie, Lundberg & Hawkins, 2012). It is within the family environment that an individual's physical, emotional and psychological development occurs. Within the family, we learn about unconditional love, understand right from wrong, as well as, gain empathy, respect and self-regulation. These qualities, according to the White Paper on Families (2013), enable us to engage positively at school, work and in society (Department of Social Development, 2013). They are the primary sources of an individual's development and considered as appropriate in society (Ogwo, 2013). While the family is considered central to the socialization of children, not much is known about the role of family functioning in the decision-making styles of adolescents.

Family functioning can be associated with better adolescent adjustment, while adolescent adjustment outcomes affect perceived family functioning (Shek, 2002). The way in which families interact or function, indicates how its members relate to one another, which activities they engage in together and the acceptance of family routines (Olson, 1993, as cited in Buswell et al., 2012). Family functioning, according to Shek (2002), generally includes the quality of family life and relates to the wellness, competence, strengths and weaknesses of a family.

As described by Olsen (1993, as cited in Buswell et al., 2012), family functioning is a delicate balance between family closeness/cohesion and family adaptability (the capacity to be flexible to the changes and challenges), both within the family and its environment (Buswell, et al., 2012). Families that emphasize personal decision-making, affect educational, occupational, recreational, marital and family choices (Zarret & Eccles, 2006). Families need to be able to generate options, weigh up those options, reach some consensus on a decision and evaluate their choices, i.e. motivate a decision-making process for effective decision-making to be reached by most of its members without dominating decision-making (Carpenter & Mulligan, 2009).

Since decision-making is a critical part of life and may determine the future success of adolescents, their families and society at large, the purpose of this study was, therefore, to examine the role of family functioning in the decision-making style of adolescents.

WESTERN CAPE

1.2. Theoretical Framework

According to Buswell et al. (2012), family functioning is regularly examined and interpreted through a family systems theoretical perspective. The McMaster Model of Family Functioning (MMFF) theory, based on a systems approach, is one variation of the family systems approach and was used to support this study. MMFF focuses on several dimensions that are considered to have the most important impact on the family (Walsh, 2003). Miller, Ryan, Keitner, Bishop & Epstein (2000) allude to the fact that a family's functioning can be evaluated according to these dimensions. These dimensions, according to Epstein, Baldwin & Bishop (1983), include problem-solving, communication, role allocation, affective responsiveness, affective involvement and behaviour control.

The MMFF views the family as an 'open system', consisting of systems within the system that relate to other systems, such as extended family, schools, industry, religions (Epstein, Baldwin & Bishop, 1983; Sheriif, Seedat & Saffla, 2010). The unique aspect of the dynamic family group cannot simply be reduced to the characteristics of the individuals or the interactions between pairs of members. There are rules, boundaries and actions by members that guide each family member's behaviour to what is considered desirable for the family (Miller et al., 2000).

1.3. Problem Statement

An important goal of adolescence is the development of decision-making skills, in light of increasing independence that would require more decisions to be made independently (Miller & Byrnes, 2001). These skills have been noted as particularly important, to set the stage for continued development through life, as adolescents begin to make their own choices and engage in a variety of activities that would influence the rest of their lives (Zarrett & Eccles, 2006). When adolescents make decisions, it is important that they have the skills to make confident ones (Deniz, 2011). Jannis & Mann (1997, in Mann, Burnett, Radford & Ford, 1997) identify three decision-making styles used to cope with stressful decisional conflicts, namely: (i) Vigilance, (ii) Hyper vigilance, and (iii) Defensive avoidance.

Factors that affect decision-making include the family environment and the relationships that individuals have with family members (Amesh, 2013). A good family environment provides close relationships, good communication and model problematic behaviour; however, a problematic environment can have the opposite affect; therefore family matters (Aufseeser, Jekielek, & Brown, 2006). There is very limited research on the role of family functioning in

the decision-making styles of adolescents. Previous studies on decision-making focus mainly on career decision-making (Martinez, 2007), consumerism (Kamaruddin & Mokhlis, 2003), academic related decisions (Miller & Byrnes, 2001) and other life satisfaction related decisions (Cenkseven-Önder, 2012). These studies do not focus on the decision-making styles of adolescence, but rather on the processes taken to make decisions and the impact they have on good or bad decision-making abilities. This study, therefore, aims to examine the role of family functioning in the decision-making styles of adolescents, as literature indicates that the family, as a whole, matters (Roman & Davids, 2013).

1.4. Research Questions

- 1. What is the family functioning of adolescents?
- 2. What is the prevalent decision-making style of adolescents?
- 3. What is the role of family functioning in the decision-making styles of adolescents?

WESTERN CAPE

1.4.1. Aim of the study

The aim of the study was to examine the role of family functioning in the decisionmaking styles of adolescents within the Overberg Area.

1.4.2. Objectives of the Study

The objectives of the study were to:

- Determine the prevalence of the dimensions of family functioning of adolescents.
- Establish the most prevalent decision-making style of adolescents.
- Examine the role of family functioning in the decision-making styles of adolescents.

1.5. Hypothesis of the study

There is a significant relationship between the role of family functioning and the decision-making styles of adolescents.

1.6. Methodology

This study was conducted within the Overberg Area, Theewaterskloof Municipality of the Western Cape. The Theewaterskloof Municipality is made up of 8 rural towns. Each of the towns has its own natural crops that play a pivotal role in the economic development of the region. This study was conducted in the Grabouw area, as part of a larger Community Engagement study, funded by the National Research Foundation, to build and strengthen the Theewaterskloof Communities. Grabouw is a mid-sized town located in the Western Cape Province, approximately 65 km south-east of Cape Town, along the N2 highway. It is located on the farther side of Sir Lowry's Pass from Somerset West, in the vast Elgin Valley, which stretches between the Hottentots-Holland, Kogelberg and Groenland Mountains, with the valley floor still being substantially hilly. Grabouw is the commercial Centre for the Elgin Valley, the largest single export fruit producing area in Southern Africa.

A quantitative research methodology was used for this study. Quantitative research, according to Creswell (2009), is an inquiry approach used to describe trends and explain the relationship between variables. It is a type of research methodology in which the researcher decides on the study, asks questions, narrows down the questions, collects numeric data, analyses these numbers using statistics and conducts the inquiry in an unbiased, objective manner. Quantitative research methodology also tests the hypotheses of a study.

1.7. Significance of the study and conclusion

As information regarding this topic is limited, this research was designed to help the investigator learn more about the perceived decision-making styles and family functioning of adolescents. Because of this information limitation, the writer believes that the outcomes of this study would also increase the knowledge of parents, teachers/schools, practitioners, adolescents and the broader society. Adolescents may realize just how important effective decision-making is and how correct or incorrect decisions could affect their lives and environment. Adolescents may also understand how their decisions impact on the family members, in the different roles that they occupy within the family system. Parents may become aware of the importance of adolescents' independent decision-making as well and help to guide their young adults in making good choices. The results may be used as a guide and framework for effective decision-making skills training programmes to be implemented for adolescents.

1.8. Definitions and descriptions of key concepts

Adolescence

Adolescence may be defined as the period in life when most of an individual's biological, cognitive, psychological, and social characteristics are changing towards becoming an adult (Ogwo, 2013).

• Decision-making

Decision-making can be defined as the process of choosing between different alternatives, while in the midst of pursuing a goal (Miller & Byrnes, 2001).

Decision-making style

A decision-making style is a process that includes the approach, reaction and action of the individual who is about to make a decision (Deniz, 2006). Decision-making styles, according to (Galotti, Ciner, Altenbaumer, Geerts, Rupp & Woulfe, 2006), are the approaches individuals take when they need to make a decision, which differs in terms of perceived effort and efficiency.

Family functioning

Family functioning includes a commitment to support the functions of the family and include: economic, safety, child rearing, care-giving and communication (Johnson, Frenn, Feetham, & Simpson, 2011). According to Shek (2002), family functioning generally refers to the quality of family life and concerns the wellness, competence, strengths and weaknesses of a family.

Problem-solving

Problem-solving can be defined as the ability to resolve problems at a level that maintains effective family functioning (Walsh, 2003).

• Communication

Communication is the manner in which information is exchanged within the family (Walsh, 2003). Effective family communication depends on clear and direct communication between family members. Expressing their feelings to one another, make family members better equipped to solve problems when they arise. Their ability to listen and pay attention to what is said is essential for effective family communication (Jobe-Shields, Buckholdt, Parra & Tillery, 2014).

Roles

Roles can be defined as the recurrent patterns of behaviour, employed by individuals, to fulfil family functions (Walsh, 2003). The establishment of these roles in a family is directly connected to its ability to deal with changes.

• Affective responsiveness

Affective responsiveness is the ability of the family to respond to a range of stimuli with the appropriate quality and quantity of feelings (Walsh, 2003). According to Jobe-Shields, Buckholdt, Parra & Tillery (2014), families need to be able to share and experience feelings such as love, tenderness, joy, fear and anger; as responding emotionally to family members, in an appropriate manner, is key to healthy family functioning.

• Affective involvement

Affective involvement is the degree to which the family, as a whole, shows interest in, and values the activities and interests of, individual family members (Walsh, 2003). Showing interest in, and valuing the activities of, other family members is essential for healthy family functioning (Jobe-Shields, Buckholdt, Parra & Tillery, 2014).

• The McMaster Model of Family Functioning (MMFF)

The McMaster Model of Family Functioning is a systems theory that identifies six features of healthy and well-functioning families: they can solve problems, communicate; have appropriate role allocation; show affective responsiveness; have empathic affective involvement; and apply flexible behaviour control (Ryan, Epstein, Keitner, Miller & Bishop, 2005).

1.9. Summary of the chapters

Chapter 1: Introduction

Chapter 1 provides an introduction to this study and describes the aims, background and rationale, theoretical framework and problem statement for the study. The objectives of the formulated research problem and hypotheses are introduced, as well as, the significance of the study. The chapter ends with the definition of concepts and the summary of the chapters.

Chapter 2: Theoretical framework

Chapter 2 provides the theoretical framework that underpins the study i.e. the Family Systems Theory, of which the McMaster Model of Family Functioning is a component. Families are described and defined, using the McMaster Model of Family Functioning, to highlight the family as the protector and outline risk factors associated with such a family system. Lastly, the family policy is discussed and its role in the functioning of families.

WESTERN CAPE

Chapter 3: Literature Review

Chapter 3 provides literature relating to the three variables that form part of this study. These variables include: adolescence; decision-making; and family functioning. The section on adolescence includes the developmental tasks associated with this phase of development, as well as, adolescents and their decision-making styles and competencies, based on their developmental stage. In the section on decision-making; the decision-making process, decision-making styles, as well as the characteristics of decision-making, is discussed. Before the closure of this chapter, family functioning and adolescence is discussed. In essence, this chapter refers to an understanding of adolescents, decision-making and the decision-making styles, together with family functioning.

Chapter 4: Methodology

Chapter 4 gives an overview of the methodology employed in this research study. It includes a discussion on the design, as well as, the sampling techniques implemented. The McMaster Model of Family Functioning and the Melbourne Decision Making Questionnaire are introduced. The procedures followed for the pilot and the main study are laid out. The data collection process, the analysis of the data and the ethical considerations that were employed are also discussed.

Chapter 5: Results

Chapter 5 provides the results that were obtained during the data collection and analysis phases of this research study. The data gathered is expressed by means of descriptive and inferential statistics. Tables outline the results obtained by the total sample.

Chapter 6: Discussion, conclusion and recommendations

Chapter 6 provides the platform for the discussion of the results that were obtained during the data analysis process. It allows for the integration of findings with literature and theory. This chapter also provides the limitations of the study, the conclusion, as well as, recommendations derived from conducting this study.

THEORETICAL FRAMEWORK

2.1. Introduction

This chapter provides the theoretical framework that underpins the study. The theoretical framework used for this study is based on the Family Systems Theory, of which the McMaster Model of Family Functioning (MMFF) is a component. The MMFF is one variation that underlies the family system model. In this chapter, families are described and defined. The MMFF is used to highlight the family as the protector and outline risk factors associated with such a family system, in relation to the six dimensions of family functioning. The family policy is included herewith as it describes the importance of how government and its policies contribute to ensuring the well-functioning of families.

UNIVERSITY of the WESTERN CAPE

2.2. Defining and describing the family

Globally, the family is perceived as the most enduring social unit, i.e. the family is central to the healthy functioning of individuals and the broader society (Hochfeld, 2007). The notion of family is valid to be a natural, inherent part of one's thinking, regarding personal life. Within society, it is perceived to be filled with symbolic significance that lies within the fluid webs of relationships. These fluid webs of relationships in society define our social family and kinship ties (Neale, 2000). The idea of kinship ties is rooted in the biological connection to ensure the order of a clan or family hierarchy through arranged marriages, patriarchal lineages; honoured by oneness of interest; and prescribed, accepted social behaviour and development (Graham & Graham, 2001). According to Makiwane, Makoae, Botsis & Vawda

(2012), the family constitutes a basic unit of relationships pertaining to reproductive processes and is defined by law or custom.

Families are the primary source of individual development and the primary setting in which children begin to acquire their beliefs, attitudes, values and behaviour considered as appropriate in society (Ogwo, 2013). Walsh (2003) describes a family as; two or more persons who live together and are interrelated either by blood, marriage or adoption. Cox & Paley (1997) also depicts the family as a hierarchically organized system, consisting of smaller sub-systems embedded within larger systems, whereby interaction occurs across all levels of generations. Sub-systems are defined by set boundaries that allow members to regulate their interaction with their environment, while still preserving the integrity of the family as a unit. They are conditioned by rules within and across these boundaries in relation to these family interactions (Cox & Paley, 1997). The White Paper on Families (2013) defines families with the most common description on families in literature: as societal groups of members related by blood (kinship), adoption, foster care or ties of marriage (extended families) including; civil, customary, religious marriages, or communal union, that extends beyond any particular shared physical residence (Department of Social Development, 2013).

According to Olson (2000), families need both stability and adaptability (also referred to as flexibility) as this will assist to distinguish functional families from dysfunctional, or at-risk, families. He further elaborates on balanced families based on the four levels of relationship adaptability. These levels of relationship adaptability include:

(1) Structured relationships indicating a level of democratic leadership. This implies stability in roles, for which there are a few rules with limited changes to them.

- (2) Flexible relationships are characterised by egalitarian leadership, in which negotiations are open and inclusive. Roles are shared, as well as, age-appropriate, and changes are applied when necessary. Unbalanced families, on the other hand, commonly entail a relationship that is either rigid or chaotic.
- (3) Rigid relationships refer to a stringent type and imply a high level of control by one member for a family. The rules are strictly defined and enforced with limited negotiations and participation in decision-making.
- (4) Lastly, a chaotic relationship is created by an unsettled, irregular or a limited type of leadership. Roles are unclear, positioned between members, and decision-making is impulsive and thoughtless.

The presentation of structure, organisation and transactional patterns within a family is seen as important variables that determine the behaviour of family members and further ensures a balance within the family (Ryan, Epstein, Keitner, Miller & Bishop, 2005). Furthermore, if these variables are in good working order, the behaviour of family members will be positively influenced. Carpenter & Mulligan (2009) add that communication is essential for successful family functioning. Families with poor communication tend to be lower functioning, whereas families with good communication tend to be higher functioning (Smith, Freeman & Zabriskie, 2009). Therefore, balanced families, as opposed to unbalanced families, tend to have more positive communication skills (Olsen, 2000). The White Paper on Families describes dysfunctional families as those in which on-going conflict, misbehaviour, neglect or abuse occurs (Department of Social Development, 2013).

According to Walsh (2003), it is important to understand the concepts of functional and dysfunctional. Functional essentially means workable and refers to the usefulness of family

patterns in achieving their goals, including instrumental tasks, such as problem solving, and the socio-emotional well-being of families that imply a goodness of fit. Dysfunctional, on the other hand, implies family patterns that are not working and that are associated with symptoms of stress. Furthermore, it must be noted that a family may be functional at one system level, yet dysfunctional on another (Walsh, 2003). The challenges experienced in one dimension of family functioning may not have an impact on another. One family may have challenges with behaviour control, another with communication, while some may have challenges with both (Epstein, Baldwin & Bishop 1983). The ineffective functioning of families, according to Ryan et al. (2005), can become a contributing factor to the emotional and physical problems of family members. Effective functioning, on the other hand, supports the optimal physical and emotional well-being of family members.

Research suggests that growing up in a well-functioning family can make a significant difference to the future outcomes of young generations (Holborn & Eddy, 2011). Dysfunctional or at-risk families can damage the desire of younger generations for prosperity. Many South African children are growing up in dysfunctional families, creating a lack of safety and security, and may further result in these children also creating dysfunctional families (Holborn & Eddy, 2011). South African children, however, may have an opportunity to prosper if they were to be given the chance to experience stability within their families and/or communities (Holborn & Eddy, 2011).

2.3. South African families

All South-African communities view family as an integral part of society, representing the key social unit (Department of Social Development, 2013; Makiwane et al., 2012). These social units are described as being open to change and characterized as a fixed structure, with

strong influences on social life (Makiwane et al., 2012). The traditional family system in the South African context comprises of related kin with mutual advantages, which holds great significance among the main cultural groups of South Africa (Amoateng, Richter, Makiwane & Rama, 2004). According to Makiwane et al. (2012), South-African families are diverse; culturally, socially and economically; consisting of nuclear, extended, re-constructed families, etc. This diversity can be seen in the structures of families: families may be extended or multi-generational; nuclear families consist of one or more parent and children; single parent with children; re-constituted families with step-parents and step-children; homosexual families; and others (Amoateng et al., 2004). In African cultures, the family is extended to include aunts, uncles, grandparents, cousins and other relatives, by blood or intermarital relations (Makiwane et al., 2012).

In South African society, the family is perceived as a group of people with companionship, sexual activity, mutual care, child bearing, shared rearing and support of children, and primarily focused on the nuclear family. In accordance to international literature this concept defines the family as a small unit, deprived of the relationship between men and women legally bound together through marriage as husband and wife (Muncie & Sapsford, 1995). This concept is, however, not true, as it is ambiguous, and is referred to as a communal family under common law, according to the South African Marriages Act No 25 of 1961.

South Africa has a series of unique situations that has affected and may still be affecting the structure and functioning of families (Holborn & Eddy, 2011). There are social stressors and historical factors that affect dynamics within the family. In addition, families are also negatively affected by social, political and economic conditions of colonialism and urbanisation (Amoateng et al., 2004). According to Makiwane et al. (2012), all these

historical changes have led to changes in the structure, responsibilities and roles in the family. It has created conditions of weakened family ties and obligations towards one another. These conditions, according to Amoateng et al. (2004), create a gap between idealization of the family and the reality, as some South Africans often find themselves living in and under conditions that compromise their beliefs regarding what families are and what families should be. However, one way of understanding how families should function is through a systems approach.

2.4. Family systems theory

Ryan et al. (2005) summarize the family systems theory by stating that families can be seen as goal directed, self-correcting, dynamic, unconnected systems which both affect, and are affected by, the environment and the qualities within the family system itself. The Family Systems Theory emerged from the General Systems Theory (Morgaine, 2001) and therefore shares the same principles. As outlined by Cox & Paley (1997), the principles of the General Systems Theory, when applied to understanding the family as an organized system, includes:

- (a) a wholeness and order, referring to the idea that a whole is greater than the sum of its parts and cannot be understood simply from the combined characteristics of each part;
- (b) an hierarchical structure, wherein systems are composed of sub-systems that are really systems of their own;
- (c) adaptive self-stabilization, referring to homeostatic features of systems that compensate for changing conditions in environments by applying coordinated changes in the internal workings of the system; and
- (d) adaptive self-organization which is complementary to the notion of self-stabilization and refers to the ability of open, living systems to adapt to change in or challenge to the existing system.

The family system is a well-established model, commonly used to take effect within the family systems framework (Buswell et al., 2012). It is commonly examined through the family systems theoretical perspective and includes:

- Family relationships function independently (Morgaine, 2001) and are governed by boundaries or explicit rules for accessing resources and support within the family (Sturge-Apple, Davies & Cummings, 2010). There are predictable patterns of interaction within the family system that assist in maintaining the families' equilibrium (Morgaine, 2001). Amoateng et al. (2004), state that the quality and level of family relationships have a direct effect on the behaviour, social competence and goal directedness of individuals. According to Bandura (2005), a family's functioning consists of interlocking relationships, rather than just the mere collection of members.
- Power refers to the messages and rules that shape members. They induce guilt, control or limit behaviour; they preserve themselves from extinction, being repetitive and abundant. They are considered to be the relationship agreements that prescribe and limit family members' behaviour over time (Morgaine, 2001). In a well-functioning family, a clear hierarchical power exists, in which individuals share power in decision-making and the younger generation have, what is referred to as, "equal overt power" (Walsh, 2003). This implies that individual choice is given, working towards a goal of family members speaking up and being respected as significant individuals who can make valuable contributions in decision-making (Walsh, 2003).
- **Structures** in a family can be created by modifications that can occur through learning, events, change or through feedback loops (Oetter & StevensDomiquez, 1998). These authors also differentiate between: first order changes, described as

minor structural changes that create vulnerability within the system, however, the system stays the same; and second order changes, which are high level changes that can cause a change within the systems structure. Feedback loops are viewed as pathways for information sharing, starting from one point of the system, through other parts and back to the point of origin. Forms of feedback include: positive feedback that plays a vital role in creating and maintaining the stability of a system; and negative feedback loops that restore and maintain the equilibrium of the system.

- Boundaries are the emotional barriers that protect and further improve the morals and values of the system (Oetter & StevensDominquez, 1998). According to Morgaine (2001), boundaries can be viewed in a series of similarities, from open to closed system, in order to establish a clear line between those inside, and those outside, the system. Open boundaries allow elements outside the family to have an influence. Closed boundaries involve the isolation of members from the outside environment; being self-contained. No family, however, as Morgaine (2001) informs, is considered to be an entirely open or closed system. According to Walsh (2003), well-functioning families possess enough strength and integrity to allow active involvement within its borders, and through their interaction, bring varied interests and excitement back into the family.
- **Family Roles**, defined by Miller et al. (2000) & Walsh (2003), are the recurrent patterns of behaviour through which family functions are to be fulfilled. They divide family roles into two areas: *instrumental* and *affective*; and further subdivide these into two more spheres: *necessary family functions*, referring to those functions that the family must be concerned with; and *other family functions* which are not necessary for effective functioning, yet arise in the life of every family.

Walsh (2003) then describes two additional and important aspects of role functioning: *role allocation*, referring to the family's patterns of assigning roles to members; and *role accountability*, referring to the procedures within the family that ensure that these assigned roles are fulfilled.

Communication patterns. Family communication is defined as the act of making information, ideas, thoughts and feelings known and refers to the *verbal* (i.e. spoken words) and *non-verbal* (i.e. facial expressions, eye contact, movement, etc.) behaviour of family members. It also assists a family to alter their cohesion and flexibility (also referred to as adaptability by Buswell et al., 2012) to meet their developmental and situational needs (Smith et al., 2009).

Family communication is a facilitating dimension for cohesion and adaptability. Cohesion, according to Buswell et al. (2012), is defined as a feeling of emotional closeness with another person. This is ultimately referred to as 'a sense of belonging'. According to Olsen (2000), emotional bonding, boundaries, coalitions, time, space, friends, decision-making, interest and recreation are variables that can be used to diagnose family cohesion. Olsen (2000) and Sturge-Apple et al. (2010) further identify four levels of cohesion: *disengaged* (very low); *separated* (low to moderate); *connected* (moderate to high); and *enmeshed* (very high levels of emotional closeness). Separated and connected levels of cohesion imply that the individuals experience a balance and are able to be both independent from, and connected to, the family. Enmeshed level implies much consensus and less independence. Disengaged level then refers to members with limited attachment or commitment to their family.

Adaptability refers to the ability of the family to change power structures, roles and rules within the relationship. This framework, therefore, implies that both family

cohesion and family adaptability are characteristics or components of healthy family functioning (Buswell et al., 2012).

Using a framework that describes the family system theory as one that focuses on family dynamics such as; relationships, power, structures, boundaries, family roles and communication patterns; family behaviour is then understood by viewing the family as a unit, instead of by its individual facets (Buswell et al., 2012).

Similarly, Morgaine (2001) identifies the following components of family systems, which include:

- (i) The family system has interrelated elements and structure. The elements are the members of the family and each element has characteristics. There are interdependent relationships among the elements and these create a structure.
- (ii) Family systems interact in patterns. These are predictable patterns of interaction to assist in maintaining the families' equilibrium and provide clues to the elements on how to function.
- (iii) Family systems have boundaries. Every system has ways to include or exclude elements, so that the line between those within the system and those outside of the system can be clear.
- (iv) Family systems function by the composition law. All the individual elements of the system results in an organic whole.
- (v) Family systems use messages and rules to shape members. Messages and rules are repetitive and redundant relationship agreements which prescribe and limit family members' behaviour.

(vi) Family systems have sub-systems. Family systems contain a number of small groups of people with relationships shared among them and are known as sub-systems, coalitions or alliances. Each sub-system has its own rules, boundaries and unique characteristics. Within the systems framework, the McMaster model of Family Functioning was developed.

2.5. The McMaster Model of Family Functioning

The McMaster Model of Family Functioning theory is based on a systems approach (Miller et al., 2000) and is known to be one variation of the family systems framework (Ryan et al., 2005). The crucial assumption of systems theory, which underlies this model, considers the following principles of a family:

- All parts of the family are interrelated;
- One part of the family cannot be understood in isolation of the rest of the family system;
- Family functioning cannot be fully understood by simply understanding each of the individual members or sub-groups;
- Family structure and organization are important factors that strongly influence and determine the behaviour of family members; and
- The transactional patterns of the family system strongly shape the behaviour of family members (Miller et al., 2000).

The McMaster model is based on the assumption that family functioning is related to the accomplishment of essential functions and tasks that can be used to distinguish between healthy and unhealthy families (Akister & Stevenson-Hinde, 1991). Specific system variables

examined are; power allocation, division of labour, patterns of inter-familial affective expression and involvement, and communication channels (Ryan et al., 2005). These assumptions appear, albeit in a refined form, in the McMaster Model of Family Functioning (Ryan et al., 2005).

Epstein et al. (1983) identify six critical areas that the McMaster approach of family functioning consists of and provide:

- (1) **Theoretical base:** The McMaster model is based on a family systems approach: the family is a system within broader systems (individual, marital) which relate to other systems (extended family, school, church/synagogue or workplace). In the McMaster model, the dynamic family group cannot be simply reduced to the characteristics of the individual or interactions between pairs of members. Clarification of the family as a system helps to identify the locus of difficulty and stress in the system, enabling therapists to identify the desired aims, objectives and methods of approach. The focus of this approach is primarily on the individual within the context of the family. It emphasises sub-systems, repeated transactional patterns, the family structure or family processes that are far removed from the immediate family or the presenting problem.
- (2) **Time frame:** The McMaster model's emphasis is placed on problems of the here-and-now, as opposed to inter-generational patterns, past origins of problems or systematic analysis of childhood issues. This implies that the importance of what is currently happening is essential to be addressed, rather than what happened in the past (Ryan et al., 2005).
- (3) **Therapist-family member relationship:** In the McMaster approach, the belief in clear and direct communication between therapist and family members, and

between family members themselves, is a very important part of the model. Since the model draws on communication theory, this emphasis is not really surprising. Clearness and directness is often liberating for family members. Family members are required to participate in open and direct problem-solving themselves (Ryan et al., 2005).

- (4) **Insight:** In the McMaster approach, insight into a problem is not a necessary or sufficient ingredient for change to occur in the system. What produces pathology in the individual is not of concern, but the process occurring within the family that produces the behaviour therapy, is. This model, therefore, helps to change the behaviour of the individual, by changing the system, however, family members have a choice to accept or decline the proposed change (Ryan et al., 2005).
- (5) Clarity of concepts and treatment: The degree of clearness in the method of treatment implies a breakdown of the McMaster model into a series of simple, discrete components that can be taught, analysed and measured. If these concepts are consistently applied, the effective success, in the majority of cases, can be assured. Clarity of concepts and methods minimizes dependence and reduces idiosyncratic approaches (Ryan et al., 2005).
- (6) **Transportability:** This is related to the degree of clarity and transporting ideas, concepts and techniques that make up any treatment approach. Unclear or inconsistent approaches may promote dependence on, rather than independence from (Ryan et al., 2005).

Based on the emphasis originally placed on having clearly defined constructs, the McMaster approach is easily transported from one setting to another.

2.5.1. Dimensions of the McMaster Model of Family Functioning

The McMaster Model of Family Functioning approach focuses on six dimensions of family functioning. These dimensions: problem-solving; communication; roles; affective responsiveness; affective involvement; and behaviour control; are considered to have the most important impact on the family (Walsh, 2003). They are further elaborated as follows:

- (i) **Problem-solving**, the first dimension, which is the family's ability to resolve problems to a level that maintains effective family functioning. Effective families solve most problems rapidly, easily and without much thought, so that, at times, there can be some difficulty eliciting and detailing the problem-solving steps they go through (Sheriif et al., 2010; Epstein et al., 1983; Walsh, 2003). The stages in this process of problem-solving are as follows:
 - Identification of the problem;
 - Communication of the problem to appropriate resources within or outside of the family;
 - Development of alternative action plans;
 - Decision regarding a suitable action;
 - Action;
 - Monitoring that action that is taken; and
 - Evaluating the success of the action.
- (ii) **Communication**, the second dimension, is defined by how the family exchanges information and includes the consideration of such variables as content, the potential for multiple messages, checking the communication sent, as appropriately conveyed and interpreted by the receiver.

- (iii) **Roles**, the third dimension, also referred to as 'role allocation' or 'family roles', is known as the repetitive patterns of behaviour that underpin the individuals' fulfilment of family functioning. The more functions that adequately fulfil and clearly allocate the house accountability mechanism, the healthier the family.
- (iv) **Affective responsiveness** is the fourth dimension and is defined as the ability to respond to a range of stimuli with appropriate quality and quantity of feelings. The more effective the families' affective responsiveness, the wider the range, and the more appropriate their responses will be, in terms of quantity and quality for the given situation.
- (v) **Affective involvement,** the fifth dimension, is defined as the degree to which the family shows interest in, and values the activities and interests of, family members.
- (vi) **Behaviour control** is the sixth dimension and refers to the pattern that the family adopts to handle the behaviour of individuals within the family. These patterns of behaviour can be seen in three specific situations:
 - physically dangerous situations,
 - situations involving the meeting and expressing of psychobiological needs and drives,
 - and situations involving socializing behaviour both inside and outside the family.

According to Miller et al. (2000), these dimensions of family functioning were formulated to create an understanding of the structure, organization and transactional patterns associated with family difficulties.

Family is a rewarding institution intended to meet the needs of both the economy and the individual's self-realisation, including autonomy (Muncie & Sapsford, 1995). Therefore, bringing a dynamic dimension to establish our understanding of family relationships, it becomes clear that relationships, commitments and responsibilities have to be worked at in order to create a functioning family. Understanding that this dynamic perspective creates fluidity and change in a gradual and on-going process, ultimately also assists in the exploration of how decision-making occurs (Neale, 2000).

2.6. Family as a caring and support network

The family is a system composed of interacting elements which, as individuals, are subjected to macro systemic influences. Throughout its lifetime, the family faces a variety of transitions that demand adaptive efforts from its members, of which adolescence will be the most important transition (Parra, Oliva, & Reina, 2013). Adolescence can, and do, open the family to a whole new array of values (Cox & Paley, 1997). Based on the changes of physical, cognitive and emotional effects experienced by adolescents; the family system becomes unbalanced (Parra et al., 2013). This can also be observed in the desire for change by one family member, constituting the family systems' need to deal with such a request (Olsen, 2000). Another more common example of changing expectations, when a child reaches adolescence, is that this stage requires more freedom, interdependence and power in the family system (Olsen, 2000).

According to Walsh (2003), families with adolescents must establish different boundaries, than families with younger children, to facilitate this transition. Boundaries between subsystems allow for differentiation in the system and must be clear, but flexible, for effective family functioning (Cox & Paley, 1997). Adequate emotional sharing; high flexibility in

rules; good levels of satisfaction of all family members; all contribute to adolescents' well-being, by reducing adolescent risk and the opportunity of dysfunctional behaviour as a pathological condition (Tafa & Baiocco, 2009). Sound family relationships consider complete involvement and effective responsiveness among all, as important protective factors (Carpenter & Mulligan, 2009).

Strong, stable and supportive families are pillars that provide the optimum framework for the well-being of children and the core foundations for them to become responsible adults (Taffa, & Baiocco, 2009). Well-being, according to Koen et al. (2011), refers to a wide range of physical and mental health attributes and also includes the social well-being of the individual. Family strengths, according to Morgaine (2001), stems from the relationship patterns of family members; their intra- and interpersonal skills, their competencies, social interaction and conditioned behaviour among family members. These strengths assist the family in creating positive identities, encouraging the development of family members' potential, enhancing the families' abilities to deal with stress and crisis situations more effectively, and lastly, contributing to the ability of the family members to be supportive of one another.

The family remains central to the lives of its members, from birth to death, and provides them with psycho-emotional and economic support that enables them to deal with difficulties, create a desire to achieve and play a critical role in society (Amoateng et al., 2004).

Through the family, care is provided until the children can assume their own responsibilities in society (Makiwane et al., 2012). The notion of care is then, according to Sevenhuijsen, Bozalek, Gouws & Minnaar-McDonald (2003), viewed as a social process, characterised by four dimensions, with a corresponding value or virtue attached to each. The first dimension is 'caring about', which represents the recognition that care is necessary (there is a need to be

met) and that the moral quality of attentiveness is required. The second dimension is 'caring for', which denotes starting the process to meet the need (after the recognition of the need) and the moral virtue of responsibility (towards addressing the need) is required. The third dimension is 'taking care of', which represents the actual practice of caring and includes a corresponding value of competence. The last dimension is 'care receiving', which requires care-receivers to acknowledge the care given and raises the moral virtue of responsiveness.

A family appears as a holistic unit in all its interactions (Department of Social Development, 2013). The basic core of the family refers to the sharing of resources, caring, responsibilities and obligations, as accepted and respected by all its members (Graham & Graham, 2001). Family is, therefore, supposed to be a mutually beneficial institution that meets the needs of the economy, society and the individual's self-identity, and later autonomy (Muncie & Sapsford, 1995)

UNIVERSITY of the

2.7. South African Family Policy

A system of explicit and institutionalized policies implies the legal recognition of the family as a social institution that contributes to social connectedness (Robila, 2011). According to Randolph & Hassan (1996), the core functions of a family can be facilitated and enhanced by a family policy. This is further elaborated upon in the next section. In defining the family, the White Paper on Families for South Africa, guided by three key strategic priorities, aimed at guiding the core functions of the family. These key strategic priorities are:

(1) **Promotion of healthy family life** focuses on the efforts preventing the breakdown of family life, by promoting positive attitudes, values and mores about the importance of strong families within communities.

- (2) **Family strengthening** is referred to as the deliberate process of giving families the necessary opportunities, relationships, networks, support and protection during adversity and social change.
- (3) **Family preservation** generally refers to keeping families together and specific programmes are intended to strengthen them during crises (Department of Social Development, 2013).

From a global perspective, families are considered to display four systematic functions (Sussman, Steinmetz & Peterson, 1999). These include: (1) intimate interdependence; (2) selective boundary maintenance; (3) an ability to adapt to change and maintain their identity over time; and (4) performance of family tasks. Family tasks, according to Sussman, Steinmetz & Peterson (1999), include:

- physical maintenance;
- socialization and education;
- control of social and sexual behaviour;
- maintenance of family morals and the motivation to perform roles inside and outside the family;
- the acquisition of mature family members by the formation of sexual partnerships;
- the acquisition of new family members through procreation or adoption; and
- the launching of young members from the family, when mature.

In a South African context, The White Paper on Families identifies four functions of families and further describes how each could benefit the individual (Department of Social Development, 2013). These include:

- (i) The formation of family and membership, by providing its members with a sense of belonging, instilling a personal and social identity, giving meaning and direction to the life to the individual.
- (ii) Provider of economic support, in that the basic needs of food, shelter, clothing and other needs required to enhance human development, is provided to the member.
- (iii) The provision of nurturance, support and socialization for the individual with the physical, psychological, social and spiritual development, values and norms.
- (iv) Fourthly, the protection of the family unit, allowing individuals to savour the feeling of being cared for, loved, welcomed and supported within the system.

As a response strategy for families in South Africa, The White Paper on Families is specifically developed through a strengths-based approach. It identifies five theoretical approaches adopted to assist in influencing the well-being of families, thereby emphasising the strengths-based approach (Department of Social Development, 2013). They are:

- (1) **The Rights Based Approach** is referred to as a conceptual framework for the purpose of human development, based on international human rights standards. It is further also directed at promoting and protecting human rights.
- (2) **The Strengths Perspective** is closely related to empowerment and seeks to identify, use, build and reinforce the existing strengths and abilities of people.
- (3) **The Cycle Approach** is also referred to as 'the family life approach'. It views family life as constantly changing. It is influenced by psychosexual development and rites of passage, such as marriage, divorce, child-rearing or retirement.
- (4) **The Systems Approach,** in which the family is viewed as a social system, its members are interdependent and any change in one family member's behaviour will ultimately cause a change in the behaviour of the rest. This approach provides

- a lens to view the family in concert with the country's history, political economy and contemporary social forces.
- (5) **The Social Development Approach,** where the family is viewed as the basic unit of society that plays a vital role in the survival, protection and development of children, implies that families need support and strengthening to assist them in meeting their goals.

The White Paper on Families directly and indirectly influences the greater well-being of most families within South Africa. Such direct influences offer support to families, and/or its members, in instances of discourse. Indirect influences relate to a family's overall access to resources, basic services, access to basic needs and rights, according to the Bill of Rights and community intervention and support (Department of Social Development, 2013). The White Paper on Families serves as a documented guideline of policy issues to Government and enables it to gauge the possible impact prior to formalisation through legislation. Bogenschneider (2006) and the Department of Social Development (2013) define family policies as government activities that are designed intentionally to support families, enhance family members' well-being and strengthen family relationships.

Levinson, Sutton & Winstead (2009) state that policy is typically characterized as a set of laws or normative guidelines, formulated according to government jargon, and is variably successful in binding communities to its mandate when implemented on a social field. Above all, policy is a creature of late modernity, the child of representative liberal democracy and represents the transubstantiation of law into a semantic register that encompasses discourses of state governing mentality (Levinson et al., 2009). Policy, basically, (a) defines reality, (b) orders behaviour, and at times, (c) allocates resources accordingly (Levinson et al., 2009). In

refined forms, a policy is the cultural-textual expression of a political practice and it makes governing statements about what can and should be done (Levinson et al., 2009).

The family policy for South Africa, drafted in the National Family Policy of 2005, emphasizes the basic assumption that the family is seen as the core of society, which includes, goals of protection and support through family strengths and opportunities, made visible through service delivery, creating sustainable families (Department of Social Development, 2013). Family functioning, according to Buswell et al. (2012), is regularly examined and interpreted through a family systems theoretical framework. All models using the systems approach have some features in common, including the ability to achieve a global perspective of families, as a change in the family system could lead to a change in the functioning of the system (Ryan et al., 2005).

Through the identification of the family as a problematic institution, the development and establishment of social welfare services and policies were regarded as necessary. The purpose of these welfare services have, in essence, primarily been the protection, support and the development of families, in line with these policies and guidelines (Muncie & Sapsford, 1995). With such policies being viewed as a formal way of social discovery, it seems to possess a sense of power to shape the way people function and think (Sevenhuijsen et al., 2003). Policy consists of contested values and preferences that, through a political process, are translated into law which, in turn, creates a power to reward, punish, encourage or prohibit, shape behaviour or exert control (Walsh, 2003). Power, as stated earlier, refers to the messages and rules that shape members and are considered to be the relationship agreements that prescribe or limit family members' behaviour, over time, within a systems approach (Morgaine, 2001). It must, therefore, be concluded that decision-making does not

happen in isolation. There are a number of elements, within the context that an individual finds him/herself, that play a role in this process (Amesh, 2013) and, ultimately, also influences decision-making in a specific time and place (Carpenter & Mulligan, 2009).

2.8. Conclusion

It can therefore be deduced that families are crucial in all societies and family functioning is important for the rearing of children within the family. Moreover, the approved, appropriate and accepted functioning of the family is central to all policy frameworks and approaches. The family operates as a system and is viewed as a structure that is constantly changing.

The purpose and value of family is widely spread and touches most of human existence. This chapter aimed to examine literature regarding family functioning, viewed in the context of the McMaster Model of Family Functioning, which included the concept of a family through a family systems approach. The Policy on Families was examined according to the values and principles that the South African Government endorses. The following chapter will give an overview of the literature, based on the variables that this research examined.

CHAPTER 3

LITERATURE REVIEW

3.1. Introduction

This chapter provides a critical insight into some of the research conducted, as well as an analytical appraisal of available literature, both internationally and locally. The literature reviewed focused, firstly, on the concept of adolescence: primarily the developmental tasks associated with this phase of development. Secondly, the concept of decision-making, the decision-making process, decision-making styles, the characteristics of decision-making, and especially, how this impacts adolescents, are also dealt with in this review. Finally, the concept of family functioning, particularly family functioning and adolescence, which is crucial to this study, is reviewed for existing literature. All these concepts are then discussed as an in-depth overview.

3.2. Adolescence and their development

Adolescence is not a static stage in life (Eccles & Gootman, 2002). According to Fantasia (2008), it is the longest developmental stage, beginning at puberty and extending into the early 20's. Louw, Van Ede & Louw (1998) asserts that it is largely characterised as starting at ages of 11-13 years and ending between the ages of 17 and 19 years. Different periods in life, according to Bandura (2005), represent different challenges and demands for successful functioning. Makiwane & Kwizera (2008) emphasize that the developmental benefits in accession are of the highest at this stage of adolescence.

According to Fantasia (2008), adolescence is a traditional stage of physical and mental human development and considered to be the most tumultuous to navigate. It is a stage characterized by rapid physical growth, reproductive maturity, psychosocial advancement and profound psychological changes (Fantasia, 2008). It involves having to learn to be more socially responsible for yourself and your actions (Kambam & Thompson, 2009). According to Louw, Van Ede & Louw (1998), this physical growth constitutes various body and hormonal development, referred to as changes.

Adolescents are considered children in some aspects and adults in others. They suffer the fate of these developmental challenges, experiences and the additional difficulties of the adult world, both as individuals and as a group (Ogwo, 2013). When analysing adolescence, it is important to consider the complex and diverse nature of adolescents' physical and sexual development, as well as, the development of their thinking, feelings, personal relationships, behaviour and identity (Louw, Van Ede & Louw, 1998). In essence, it can be said that, ultimately, adolescence can be defined as a stage in life when most of an individual's biological, cognitive, psychological and social characteristics are challenged with childlike perceptions contrasted with those of adults (Ogwo, 2013). It is in this period in life that they establish their own independence (Walsh, 2003).

The word 'development' refers to all changes experienced by individuals throughout their lifetime (Louw, Van Ede & Louw, 1998). Not all changes are considered in developmental psychology, but the ones that do, relate to:

- (i) changes based on relative permanency;
- (ii) changes that are connected to other important changes; and
- (iii) changes that link to other developmental patterns.

Developmental stages refer to the dividing of these changes into categories pertaining to the individual's experiences, from conception to death. The characteristics and skills associated with the individual, at a particular stage of development, are considered to be the developmental tasks (Louw, Van Ede & Louw, 1998).

The developmental tasks associated with this stage, according to Louw, Van Ede & Louw (1998), includes the following:

WESTERN CAPE

- the acceptance of bodily changes;
- the acceptance of, and adjustment to, certain groups;
- the development of gender role identity;
- the development of independence;
- the development of strong emotional bonding;
- the development of socially responsible behaviour;
- the development of moral concepts and values;
- the development of a value system and philosophy of life;
- the selection of, and preparation for, a career;
- the establishment of heterosexual relationships; and
- the achievement of financial independence.

Eccles & Gootman (2002) summarise these developmental tasks of adolescence as:

- the shifts in relationships with parents from dependency and subordination to an increasing maturity and responsibility in the family;
- the exploration of new roles;
- the experience of intimate partnerships;

- identity formation at both the social and personal levels;
- planning the future and taking the necessary steps to pursue those plans; and
- acquiring the skills and values needed for a successful transition into adulthood.

Adolescence is also often considered to be a time of increased impulsivity and thrill seeking behaviour (Kambam & Thompson, 2009). Impulsivity is described as diminished self-control, or response inhibition, that leads to impetuous behaviour and may involve aspects of the cognitive control system (Kambam & Thompson, 2009). Thrill-seeking behaviour, on the other hand, is perceived as the willingness to take risks, in order to seek out stimulating or novel experiences. This behaviour may involve aspects of the socio-emotional system.

This stage represents the developmental phase where critical decision-making occurs on the most important key life transitions (Makiwane & Kwizera, 2008). In addition, the most important cognitive changes considered for this period in life, relate to the individual's ability to think theoretically, consider the hypothetical, as well as the real, process information in a complex and more detailed manner, consider the multiple dimensions of a problem and have the ability to reflect on the self and on complicated situations (Eccles & Gootman, 2002). According to Louw, Van Ede & Louw (1998), cognitive changes lead to a transformation from concrete thinking to more comprehensive thinking. This implies that an adolescent's ability to analyse and reason more logically becomes advanced. As adolescents gain further experiences in navigating their social world, they begin to better understand their own behaviour and tendencies, thereby creating an ability to more accurately predict or presumably control their decision-making (Albert & Steinberg, 2011).

Adolescence is a stage of improved: logical reasoning; short and long term memory; efficiency of, and capacity for, processing information (Kambam & Thompson, 2009). These

decisions involve: education; entering the job market; adopting healthy lifestyles; starting a family; and participating in society (Makiwane & Kwizera, 2008). Adolescents' cognitive development contributes to the formation of their identities and, along with the ability to think abstractly and follow a decision-making process, aids them to resolve choices regarding their value system and roles (Louw, Van Ede & Louw, 1998).

Eccles & Gootman (2002) identify some primary challenges that emerge, when adolescents begin to take on the above developmental tasks, as follows:

- the management of these demanding tasks;
- identifying personal strengths and weaknesses;
- refining individual skills to coordinate and succeed in these tasks;
- finding meaning and purpose in these tasks; and
- assessing and making the necessary life changes to cope with these tasks.

Zarrett & Eccles (2006) further highlight that the successful management of these tasks (also referred to as challenges) depends on: the psychological, physical and cognitive assets of the individual; the support available; and the developmental settings in which adolescence can explore and interact with these challenges. In addition, personal assets believed to be critical for healthy development during this period include:

- having confidence in the ability to achieve goals and make a difference;
- having a strong desire to engage in important activities;
- mastering learning tasks;
- becoming socially connected;
- developing the ability to control and regulate emotions;
- have a sense of optimism; and

developing a connection to, or commitment in, customary pro-social institutions
 e.g. families

(Zarret & Eccles, 2006).

The fact that all these changes occur during this phase, reinforces the notion that adolescence is most likely to be a period of significant high risk (Kambam & Thompson, 2009). Autonomy (in this case, meaning the ability to make decisions), according to Frank, Pirsch & Wright (1990), is often the first qualification to be associated with adolescence. It is also assumed to be a key developmental task during adolescence in western societies (Van Pentegem, Beyers, Vansteenkiste & Soenens, 2012).

3.2.1. Adolescence and autonomy

In adolescent developmental psychology, autonomy is frequently defined as an independence or self-reliance, which generally refers to how an individual behaves or decides without relying on others (Van Pentegem et al., 2012). According to Louw et al. (1998), autonomy in adolescents refers to their ability to make independent choices and assume responsibility for their choices. Although the development of autonomy represents a crucial developmental task for adolescents, the conceptual and empirical distinction between the two definitions of autonomy is important, each referring to different processes. In interpersonal terms: (1) Autonomy as independence versus dependence refers to what degree adolescents decide, act, think without relying on others. Autonomy as independence is then in contrast to autonomy as dependence; (2) Autonomy as self-endorsed functioning is the experience of the individual that accompanies the dependent or independent behaviour. This refers to what extent they act on their own values, goals and interest, whether independently or with dependence on others (Van Pentegem et.al, 2012).

Smetana, Crean & Campione-Barr (2005) suggest four domains that govern decisions, depending on the level of autonomy (autonomy being independence and self-reliance). These domains are as follows:

- (1) **personal domains**, such as appearance, where decisions only impact the individual and pertain to the private sectors of life;
- (2) **social conventional domains,** where cultural norms within the family or society serve as guidelines;
- (3) **prudential domains** that have potential negative consequences for health and safety; and
- (4) **multi-faceted domains** that reflect overlapping domains and typically traverse the personal and conventional or prudential domains.

Depending on the degree of the internalization of the motives underlying behaviour, decision-making will be enacted either with a sense of freedom, choice and volition or instead with a sense of pressure, control and coercion. Autonomy is then ultimately perceived as a voluntary or sanctioned functioning, rather than a pressured or controlled functioning (Van Pentegem et al., 2012).

Decision-making is one of life's most significant activities (Jurisova & Sermany-Schuller, 2013). Having the ability to make wise, intellectual decisions is considered fundamental to living a prosperous and accomplished life (Armesh, 2013). The quality and results of decisions could significantly influence the effectiveness of the individuals personal functioning (Jurisova & Sermany-Schuller, 2013). Ultimately, adolescents who are able to make effective decisions are more content with their environment and their satisfaction is enhanced (Cenkseven-Onder, 2012).

3.3. Decision-making

Decision-making involves a process of identifying a course of action that is aimed at solving a problem. According to Miller & Byrnes (2001), decision-making can be defined as the process of choosing between different alternatives, while in the midst of pursuing a goal. Decisions are regarded as choices made, which influence the environs to an extent that depends on how the desired goals are met. Hence, the optimality of a decision may vary as a function of accuracy of the judgment on which it is based (Jackson & Kleitman, 2014). Effective decision-making is, therefore, considered to be an important life skill (Roman & Davids, 2013).

Decision-making includes a series of cognitive operations that form part of the cognitive process (Miller & Byrnes, 2001). Deniz (2006) defines a cognitive process as an inclination to one of several available choices, in order to make an effective and healthy decision. Jackson & Kleitman (2014), add that an engagement into a decision-making process is required to facilitate the attainment of such goals. Van Pentegem et al. (2012) conducted a study that focused on the concept of adolescent autonomy (the ability to make decisions) and its association with the psychosocial functioning of the individual. The aim of this study was to differentiate between: (a) autonomy defined as independence versus dependence and (b) autonomy defined as self-endorsed versus controlled functioning. The findings indicate that the degree of independent decision-making could clearly be differentiated from the underlying motives for doing so. Independent decision-making showed unique associations with more problem behaviour. Self-endorsed motives, for both independent and dependent decision-making, generally related to an adoptive pattern of psychosocial functioning, and controlled motives were associated with maladjustment.

According to Fantasia (2008), the term 'decision-making' has many surrogate terms, often referred to as critical thinking, reflective judgment and problem-solving. Critical thinking involves an act of thorough exploration of a complex problem (Armesh, 2013). Judgments, according to Jackson & Kleitman (2014), are beliefs or predictions that may vary in the extent to which they are accurate reflections of reality. Kambam & Thompson (2009) define judgement as a process of attaching a variety of importance to several possible consequences of decisions. Problem-solving is defined as the process of taking action to meet objectives (Armesh, 2013). These terms, according to Fantasia (2008), are all considered components of the decision-making process through which an end result can be achieved.

3.3.1. The decision making process

The process of making a decision is undoubtedly complex (Kambam & Thompson, 2009) and must be regarded as a continuous sequence integrated within our interaction with people (Armesh, 2013). The Cognitive abilities developed, equips the individual with an ability to process information, reason about the steps to be taken, and resolve the difficulties that may emerge during the decision-making process (Lizarraga, Baquendano, Oliver & Closas, 2009). Similarly, Kambam & Thompson (2009) concur that cognitive capacities shape the decision-making process and psychosocial immaturity may affect decision-making outcomes in more subtle, but not necessarily less important, ways. Cognition may modify aspects of the affective responses and emotion may modify the cognitive processes, influencing values and preferences that impact cost-effective analysis when making a choice. Mature decision-making therefore includes components of: development of control over behaviour and emotions; initiation; persistence; following the processes of achieving goals; and the navigation of complex social situations (Kambam & Thompson, 2009).

Burnett (1991) describes seven procedural criteria of Janis & Mann (1977, in Mann et al., 1997) for decision-making as follows:

- (1) thoroughly canvas a wide range of alternative courses of action;
- (2) survey the full range of objectives to be fulfilled and the values implicated by the choice;
- (3) carefully weigh the costs and risks of both the negative and positive consequences which could flow from each alternative;
- (4) intensively search for new information relevant to further evaluation of the alternatives;
- (5) correctly assimilate and take account of any new information to which one is exposed;
- (6) re-examine the positive and negative consequences of all known alternatives before making a final choice and;
- (7) make detailed provisions for implementing the chosen course of action, with special attention given to contingency plans that might be required, if various risks were to appear.

Armesh (2013) also highlights that a decision-making process typically has five steps, summarized as follows:

(i) The first step is an awareness of the problem or opportunity that requires surveillance of the internal and external environment for issues that merit executive attention. Ultimately, this step refers to a process of searching for decision-making opportunities, rather than focusing primarily on identifying the opportunities based on the problem.

- (ii) Once the problem or opportunity has been recognized and analysed, decision-makers commence action and generate possible alternative solutions that will respond to the needs of the situation and correct the underlying causes. The choices decided on are alternatives and the decision-maker chooses the best alternative of those considered.
- (iii) The next step in the process is to evaluate the alternatives. The overall importance of the pros and cons relating to each alternative is important and the expected payoff associated with each should also be considered.
- (iv) Once feasible alternatives have been developed, the best alternative must be chosen and implemented. The ultimate success of the chosen alternative depends on whether it can be translated into action.
- (v) Decision-makers then gather information about how well the decision was implemented, and whether it was effective in achieving the desired goals.

UNIVERSITY of the WESTERN CAPE

Burnett (1991) conducted a study that investigated the influence of the self-concept on decision-making behaviour. The results suggest that a positive decision-making self-image is associated with using productive decision-making criteria, while a negative decision-making self-image is associated with not following prolific procedural criteria for decision-making. Life is about making decisions, but how those decisions are made has definite implications across all facets of life (Brown, Abdallah & Ng, 2011).

3.3.2. Decision-making styles

Galotti et al. (2006) defines decision-making styles as a subset of broader cognitive styles, which implies, the manner in which intellectual abilities, and an approach to cognitive tasks, are employed. Janis & Mann (1977, in Mann et al., 1997) identified a

few decision-making styles used by individuals to cope with stressful decisional conflicts. These are:

- Vigilance: The vigilance decision-making style comes into effect when an
 individual is optimistic about finding an alternative solution or answer to a
 decisional conflict. S/he also believes that there is time to follow the steps in
 order to make a good choice.
- Hyper vigilance: The hyper vigilance style comes into effect when the
 individual is optimistic about the alternatives to the decisional conflict, but
 does not believe that there is sufficient time to search for a possible
 alternative or solution. Panic and stress develops and an objectionable
 alternative is considered.
- **Defensive avoidance:** The defensive avoidance style comes into effect when the individual is pessimistic about the alternative or solution to a decisional conflict and adopts the **procrastination decision-making style**—delays or postpones making the decision; or the **buck-passing decision-making style**—the responsibility is passed on to someone else (Burnett, 1991).

Deniz (2006) conducted a study on decision-making styles and life satisfaction. The findings showed that life satisfaction is significantly correlated to all decision-making styles (vigilant, hyper vigilant, procrastination and buck-passing). Life satisfaction was found to be positively associated with the vigilant decision-making style, while negatively associated with the hyper vigilant, procrastination and buck passing styles. The study further suggests that the family environment has to be a supportive environment and that life satisfaction refers to a level of quality perceived based on the individual's personal criteria (Deniz, 2006).

Steinberg, 2005 examined the ways in which emotional and cognitive factors influence the decision-making of adolescents. Their findings indicate that adolescents are relatively more approach orientated in response to positive feedback and less avoidant in response to negative feedback. Similarly, Cenkseven-Onder (2012) examined decision-making styles among adolescents as a predictor of adolescence family environment satisfaction. He found that the complacency style of decision-making, adolescents' need for independence and the inability to make their own decisions to satisfy their needs, may diminish the satisfaction they perceive from their family. Furthermore, with the exception of the panic style, decision-making styles are predictors of life satisfaction. Ultimately, adolescents who make effective decisions are more content with their environment and enjoy increased satisfaction.

Roman & Davids (2013) examined the decision-making styles of youth in South Africa with a sample of university students. Their findings revealed differences in the decision-making styles of students living on campus, as opposed to those living off campus with their families, because the decisions of students living on campus were more controlled and self-centred, as they lived alone with no family members close by to assist with decision-making.

3.3.3. Decision-making and adolescents

An important goal of adolescence is the development of decision-making skills, especially as adolescents desire more independence and autonomy to make decisions on their own (Miller & Byrnes, 2001). Decision-making capacities of adolescents cannot be viewed in a one-dimensional framework (Kambam & Thompson, 2009). Full transition through this phase should result in the ability to: draw conclusions from available information and predict possible outcomes, reason theoretically and

understand more abstract concepts, such as love and values (Zarrett & Eccles, 2006). Cognitive skills developed over this period also enables adolescents to manage their own learning and problem-solving, while also facilitating their identity formation and maturation of moral reasoning (Lizarraga et al., 2009).

According to Steinberg (2005), the adolescent's decision-making capability improves with maturity and develops more competence. There is a distinct increase in the adolescent's capacity to think abstractly, consider multiple dimensions of problems, process information and stimuli more efficiently, and reflect on the self and life experiences (Zarrett & Eccles, 2006). Decision-making competence, according to Seiffge-Krenke (2002), may have far reaching consequences for many of life's contexts for adolescents. These decisions have long-term impact, not only for the individual, but also for their families, their communities and the economy (Makiwane & Kwizera, 2008). According to Albert & Steinberg (2011), adolescents make choices that reflect logical normative evaluation of loss and gain probabilities. They further suggest that the adolescent's lack of experience with negative consequences, support mature, categorical avoidance of choices and show a conscious evaluation of the loss and benefits of behaviour.

Kambam & Thompson (2009), refer to adolescents' limited decision-making capacities as common knowledge for centuries. Typical adolescents have deficiencies in their decision-making abilities as a result of a deficiency in their cognitive and psychosocial characteristics related to their level of development. These characteristics include: their sense of responsibility (identity); temperance (risk-taking); and perspective (future time perspective). Adolescents are particularly susceptible to the potentially harmful effects

of emotions on decision-making. According to Lizarraga et al. (2009), youth often make decisions under the sway of emotions and social aspects. However, decision-making in adolescence is influenced by other variables as well (Kambam & Thompson, 2009). Decision-making can be influenced by family, friends and social institutions (Brown, Abdallah & Ng, 2011). Other factors include the family environment and the relationships that individuals have with family members (Amesh, 2013).

3.3.4. Relational aspects of decision-making

Knowledge about the factors that affect decision-making is important as it facilitates the process of making efficient and appropriate decisions (Byrnes 1998, as cited in Lizarraga et al., 2009). Decision-making does not happen in isolation. There are a number of elements, within the context that an individual finds him/herself in, that play a role in decision-making (Amesh, 2013), ultimately, also influencing decision-making in a specific time and place (Carpenter & Mulligan, 2009). It can, therefore, be concluded that the effectiveness and management of decision-making capabilities may have been taught at home (Roman & Davids, 2013).

The theoretical underpinnings that draw the correlation between family functioning and decision-making are derived mainly from the general family literature. The family is considered central to the socialization of children (Department of Social Development, 2013), yet, as mentioned in previous sections, not much is known about the role of family functioning in the decision-making styles of adolescents. However, families that emphasize personal decision-making, influence educational, occupational, recreational, marital and family choices (Zarret & Eccles, 2006). Decisions about aspects of youth and family life are considered important indicators of the family process that contributes to later development. Decision-making is also considered part of the set of

family processes. The processes whereby parents and children negotiate decisions are characterized as "autonomy granting" or "independence giving". This process reflects a normative assumption of the right and power to make decisions, initially being the responsibility of the parent, who later transfer this decision-making ability to the children in a semi-controlled manner (Romich, Lundberg & Tsang, 2009).

3.4. The influence of family functioning

Family functioning is dynamic (Ryan et al., 2005) and refers generally to the quality of family life concerned with the wellness, competence, strengths and weaknesses of family members (Shek, 2002). A family's functioning is likely to vary according to the family's stage of development, illness, change in family roles and the changing composition of any individual family member (Ryan et al., 2005). According to Stattin, Persson, Burk & Kerr (2011), family functioning may also change over time in some families. Well-functioning families, however, create and sustain positive and sustainable value systems, through the socialization of children and the inhibition or promotion of certain behaviours among family members (Amaoteng et al., 2004).

Stattin et al. (2011) conducted a longitudinal cluster analysis study that highlighted changes in a family over time. They hypothesised that, when family functioning changes, adolescents imagine that influence and parents knowledge of their whereabouts will change too. The analysis included three family functioning measures, which are: (i) adolescent openness; (ii) parental harsh treatment; and (iii) parental openness. Two groups showed improved family functioning (increased adolescent openness in conjunction with increased parental openness, decreased parental harsh treatment, or both). Two groups showed deteriorating family functioning (decreased adolescent and parental openness, and increased parental harsh

treatment). One group showed worse parental behaviour (decreased parental openness) and one other group showed improved parental behaviour (decreased parental harsh treatment). Three groups did not display significant changes in any of the family functioning measures. However, 8:63% of the participants in clusters showed significant changes, which support their view of the family as a system.

Functional families, according to Saunders (1999), contribute to societal stability on three levels as follows:

- (1) The first level is the **micro level**, referring to the stability of the individual's personality. Families, however, still provide emotional gratification and play a crucial role in the socialization of children;
- (2) The second level is the **messo level** and refers to the stability of the family unit itself that continues with the various functions like developing strategies to live together, sharing domestic tasks (roles), resolving conflict (communication) and adjusting to the changes over time;
- (3) The third level is the **macro level** which refers to the stability of the relationship between the family and other social institutions.

Dysfunctional families, according to Department of Social Development (2013), display conflict, misbehaviour, neglect, or abuse: occurring continually or regularly.

3.5. Family functioning and adolescence

Family factors such as divorce, single parent families, family disintegration, lack of involvement, poor communication can influence the adolescent's attitude and behaviour (Louw, Van Ede & Louw, 1998). From an ecological perspective, individual characteristics and the context of the family assist in shaping youth development (Wray-Lake, Crouter &

McHale, 2010). Family functioning has been acknowledged to have an impact on the developing adolescent. The role of the family, therefore, seems crucial in adolescent development (Koen et al., 2011). Considering that the family is seen as a working unit that is greater than the sum of its parts, each individual not only affects the family as a whole, but also each individual member of that family. Characterised by a process of co-evolution, parents and adolescents should engage together in these developmental challenges with both parties being obliged to take on different tasks (Tafa & Baiocco, 2009).

The way in which the family interacts or functions, indicates the way in which the family, as a whole, relates to each member and includes activities that the family engages in together and the acceptance of family routines (Zarret & Eccles, 2006). Research confirms that good family relationships – adequate emotional sharing, high flexibility in rules, good levels of satisfaction of all family members – are important protective factors that contribute to the well-being of adolescents (Tafa & Baiocco, 2009). Shek (2002) examined the association between family functioning and adolescent adjustment. The results show that family functioning was significantly related to adolescents' psychological well-being, i.e. existential well-being, life satisfaction, self-esteem, and a sense of mastery. The results further indicate that adolescents, who perceived the family as dysfunctional, displayed more problem behaviour in both poor and non-poor groups.

Henderson, Dakof, Schwartz & Liddle (2006) investigated the direct and indirect effects of family functioning on the severity of adolescent externalizing problems. Their findings indicated that family functioning exerted direct and shared effects on externalizing problems. It is within the family environment that an individual's physical, emotional and psychological development takes place. In addition, the family provides a foundation for: learning

unconditional love; understanding right from wrong; as well as gaining empathy, respect and self-regulation. These qualities create the ability to engage positively in all domains of life (Department of Social Development, 2013).

Similarly, Parra, Oliva & Reina (2013) conducted a longitudinal study with a group of adolescents. In this study the relationship between adolescents and their parents were analysed over a period of 10 years. The results indicate that the levels of communication and affection diminished, however, their perceived adaptability remained and cohesion increased as the frequency of conflict increased. Continuous and unresolved conflict may result in serious self-esteem, autonomy and self-control deficiencies in adolescence (Koen et al., 2011).

In a study by Romich et al. (2009), the authors investigated whether adolescents make decisions autonomously, share decisions with their parents or have decisions made for them by their parents. They also examined how childhood behaviour and competence influence decision patterns in adolescence. In both models used in this study (the individual and sibling effects models), those with higher verbal ability shared more decision-making with their parents. Those who were impulsive, more than likely, made decisions autonomously. The results also suggested that children, directly and indirectly, influenced house family sharing patterns.

Fostering adolescents' self-endorsed functioning, through empathy, giving choice wherever possible, and encouraging them to act upon their personal values and interests, seems to be crucial in order to deal successfully with the challenges of raising an adolescent (Van Pentegem et al., 2012). A family, according to Stattin et al. (2011), should maximize adolescents' willingness to talk to their parents about their thoughts, feelings and activities.

As adolescents mature, they often seek more independence and autonomy and may begin to question family rules and roles, leading to conflicts of various kinds (Eccles & Gootman, 2002). However, this conflict, as found by Eccles & Gootman (2002), create a distancing in the parent-adolescent relationship, which may have great functional value for the adolescent. It may foster the adolescents' individualisation from their parents, allowing them to try more things on their own and develop their own competence and efficacy (Eccles & Gootman, 2002). Families need to be able to make choices and not dominate decision-making (Carpenter & Mulligan, 2009).

3.6. Conclusion

Literature clearly regards adolescence as a unique developmental phase in life which constitutes many challenges. The decision-making of adolescents is regarded as one of the key areas of development that determines their ability to succeed in life. Family functioning is also regarded as a critical part of an adolescent's life, as the family is where the individual begins to acquire the beliefs and skills to deal with challenges. The following chapter provides insight into the methodology used in this study.

CHAPTER 4

METHODOLOGY

4.1. Introduction

This chapter provides an outline of the methodological process followed in this study. Methodology is the process, framework and design used in an endeavour to gain the knowledge required to answer the research questions. It further enables the researcher to follow a reliable and valid process for the collecting and analysing of required data. The aim of this study was to examine the role of family functioning in the decision-making styles of adolescents, using a quantitative research methodology. The variables of this study included family functioning, decision-making and adolescence. This chapter will also provide an overall view on the course of action taken in this research, starting with the request for ethical clearance to conduct the research (for the pilot and the main study) through to the collection and, finally, the analysis of the data.

4.2. Objectives of the study

The objectives of this study were to:

- Determine the prevalence of the dimensions of family functioning of adolescents;
- Establish the most prevalent decision-making style of adolescents; and
- Examine the role of family functioning in decision-making styles of adolescents.

4.3. Research methodology

This research study used a quantitative research methodology. Quantitative research, according to Creswell (2008), is an inquiry approach, useful for describing trends and explaining the relationship among variables. It is a type of research, in which the researcher decides what to study, asks questions, narrows questions, collects numerical data, analyses these numbers using statistics and conducts the inquiry in an unbiased, objective manner. Quantitative methodology also tests the hypotheses of a study.

4.4. Research design

A research design is the plan and procedure for research that spans the decisions, from broad assumptions to detailed methods of data collection and analysis. The selection of a research design is also based on the nature of the research problems, the researcher's personal experiences and the audiences for the study (Creswell, 2008). This study used a cross-sectional, correlational research design. A cross-sectional study observes a phenomenon at a particular period of time (Gray, 2009), and one advantage thereof is that it can reveal associations among variables. Correlation research addresses questions about the relationship between two or more variables and the extent to which they co-vary (Arthur, Waring, Coe & Hedges, 2012). When an association is measured numerically, a correlation coefficient is obtained that gives the strength and direction of the relationship between these variables (Gray, 2009). Research variables are concepts that vary in ways that can be observed, recorded and measured (Miley, O'Melia & DuBois, 2009). Using the cross-sectional, correlation research design allowed the researcher to reveal the associations between these variables.

4.5. Study population

This study formed part of a larger research project, in the Faculty of Community and Health Sciences, based in Theewaterskloof. The focus of the main project was building communities, and this study formed part of project five, of six smaller projects, within the larger project, focusing on the decision-making styles of adolescents. The current study was within the first phase of project five, which is a baseline assessment of psychosocial effects on the decision-making styles of adolescents. Specifically, this study only examined the role of family functioning on the decision-making styles of adolescents in the Grabouw area.

The population refers to the group (of people) about whom researchers want to draw conclusions (Babbie & Mouton, 2010). The population of this research study comprised of 2,571 Grade 9-learners in the Grabouw area of the Western Cape Education Department. The choice of using Grade 9-learners was based on the fact that learners in Grade 9 were expected to choose their subjects for Grades 10 to 12. This meant that they were expected to make an important life decision. How they made this decision would, therefore, play a very important role in their individual lives. The total number of schools in the Grabouw area was three. This current study took place within the Theewaterskloof Municipality, in the Overberg Region of the Western Cape. The Theewaterskloof Municipality is made up of 8 rural towns. Each of these towns has their own natural crops that play a pivotal role in the economic development of the region. This study was conducted in the Grabouw area, as part of a larger Community Engagement study, funded by the National Research Foundation, to build and strengthen the Theewaterskloof Communities. Grabouw is a mid-sized town, located in the Western Cape Province, approximately 65 km south-east of Cape Town, along the N2 highway. It is located on the farther side of Sir Lowry's Pass from Somerset West, in the vast Elgin Valley, which stretches between the Hottentots-Holland, Kogelberg and Groenland Mountains, with the

valley floor still being substantially hilly. Grabouw is the commercial Centre for the Elgin Valley, the largest single export fruit producing area in Southern Africa.

4.6. Study Sample

A sample is a small selection from a population, and in statistics; the sample is a small selection of all the scores on a variable (Howitt & Cramer, 2000). In this study simple random sampling was used. Simple random sampling requires a sampling frame of all the members of the desired population, from which a sample is taken using a random sampling method (Howitt & Cramer, 2000). A sampling frame is merely a list of all members of the population, each member being allocated a number, with the appropriate size sample being selected by generating random numbers (Howitt & Cramer, 2000). All grade 9 learners of the three schools within this area were identified as participants. Contact was made with these schools and consequently, the parents and learners, with the assistance of the teachers, were requested to complete consent forms, after being fully informed about the study, in order to participate. The total sample of the study consisted of (N=243 [53.8%]) female and (N=209 [46.2%]) male participants. Participants identified their race of which, (N=248 [56.4%]) was coloured, (N=186 [2.3%]) black African, and (N=6 [1.4%]) white. The majority of participants was Afrikaans speaking (N=253 [56.5%]) followed by isiXhosa (N=166 [37.1%]) and English (N=3 [0.7%]). Other languages illustrated a total of (N=26 [5.8%]).

The Yamane formula applied to assist in computing the sample size states:

$$n = \frac{1 + N e^2}{}$$

n = sample

N = population

 e^2 = probability error

(Yamane, 1967)

The total sample size computed for the study, using the Yamane formula, was 350 Grade 9 high school learners, both male and female, from the Overberg Educational District, Theewaterskloof, within the Western Cape.

4.7. Instruments

Self-reported questionnaires were used to collect the required data from the participants. The questionnaire consisted of three sections: (i) Demographic details, such as age, gender, race, home language and family structure; (ii) The Family assessment device, based on the six dimensions of the McMaster Model of Family Functioning, identifying the level of participants' family functioning; and (iii) the Melbourne Decision-making Questionnaire that assisted in identifying the decision-making styles of participants. The instruments were translated from English to Afrikaans by a registered language specialist. The Afrikaans version was then translated back into English by a different specialist. Xhosa translation was not deemed highly necessary as English is regarded as the main South African language and Afrikaans the second highest language. Furthermore it was noted that the African scholars having being Xhosa speaking also speaks Afrikaans and/or English as an additional language it was seen fit not to translate the questionnaire into a Xhosa version.

4.7.1. The Family Assessment Device

The Family Assessment Device was designed to assess the dimensions of the McMaster Model according to family member's perceptions (Ryan et al., 2005). It consists of

subscales assessing the six dimensions of the McMaster model, as well as, a general functioning scale, which assesses the overall level of family functioning. The McMaster Model of Family Functioning is one of the variations of the Family Systems Model. One of the basic tenets of the McMaster approach is the emphasis on constructs and procedures, which can be empirically measured and verified. After developing the models of family functioning, empirical measurement of the constructs of the model was done. Thereafter, the development of a self-report questionnaire began, being The Family Assessment Device (FAD). The FAD consists of 60 statements, which family members scrutinize, then rate how well each statement describes their family, by selecting one of four responses: strongly agree, agree, disagree and strongly disagree. The FAD is scored by summing up the endorsed responses (1-4) for each subscale, which is then divided by the number of items in each scale. The scale score ranges from 1.0 (best functioning) to 4.0 (worse functioning). The questionnaire is completed by participants over the age of 12 years and the family scores are obtained by averaging the ratings of the participants on each dimension.

4.7.2. Melbourne Decision-Making Questionnaire

The Melbourne Decision-Making Questionnaire is based on the conflict theory by Janis & Mann (1977, in Mann et al., 1997). The Melbourne Decision-Making Questionnaire is a 22 item questionnaire that identifies the factor assessment of self as decision-maker in the sense of effective decision-making ability (Mann, Burnett, Radford & Ford, 1997). In this study, two decision making-styles, the hyper vigilance and vigilance decision-making styles, are selected from the Janis & Mann (1977, in Mann et al., 1997) coping conflict model of decision-making styles. The other two styles are the buck-passing and procrastination decision-making styles. The vigilance decision-making style is considered adaptive and the other three decision-making styles are

considered mal-adaptive. Grading was done by giving numerical values to items according to the answers, such as: true for me - score 2; sometimes true - score 1; not true for me - score 0. Higher scores indicated a higher level of decision-making.

4.8. Data collection for the pilot study

The research proposal was submitted to the University of the Western Cape's Higher Degree's Committee for ethics clearance. Once ethics clearance was received, the Western Cape Education Department was contacted for permission to conduct research in the schools identified. The respective principals of the selected schools were contacted and the research team met with the principal and the teachers to inform them of the study. Information sheets (Appendix IV) about the study were left for the parents of the high school learners, as well as Informed consent forms (Appendix II) that needed to be completed by the parents. Once the Informed consent forms were collected by the teachers, a date and time was set for the research to be conducted at the selected schools.

Fifteen per cent of the sample was used in a pilot study to assess the reliability of the particular instrument. The participants were informed of the study and allowed to ask questions or raise concerns that they might have. The learners were given Assent forms (Appendix III) that needed to be signed before conducting the study, with voluntary participation. They were also informed of their right to withdraw from the study, at any time, without prejudice. The questionnaires were then handed out to the intended sample for the pilot study; thereafter the questionnaires were handed out to a second group of participants who formed part of the pilot study, to allow for a test-retest method of the chosen questionnaire / instrument being the Melbourne Decision Making Questionnaire and the Family assessment device for the main study. The test-retest method was used to assist in

measuring the internal consistency of the questionnaire and establish any challenges or limitations that might have occurred. The dispensing of the questionnaire was, as far as possible, conducted with groups of 30 to 40 learners at a time, in a class room setting within the limitations of the school time table. The pilot study measured the reliability of the instrument, explored language options in the questionnaire and the consideration of the data collection process. After the pilot study, the research participants for the main study were selected. The questionnaires were completed in approximately 45 - 60 minutes.

4.9. Results

The participants highlighted the items that they were challenged with. The Cronbach Alpha scores obtained from the pilot study indicated that the Flinders Decision-Making Questionnaire had the lowest alpha score (see Table: 4.1.).

Table: 4.1. Cronbach Alpha Scores for the pilot study

Scale	Alpha Score (α)
Parental Style and Dimension Questionnaire	.90
Flinders Decision-Making Questionnaire	.58

Based on the low alpha score of the Flinders Decision-Making Questionnaire in the pilot study, it was replaced with the Melbourne Decision-Making Questionnaire, which is a revised version of the Flinders Decision-Making Questionnaire, with improved internal consistency and reliability.

4.10. Data collection for the main study

Collecting data, according to Creswell (2009), means identifying and selecting participants for a study, obtaining their permission to conduct the study and gathering information by asking questions or observing behaviour. Once the pilot study was completed, i.e. both the data collection and the analyses of the research process and the questionnaire, the results of this process informed the process for the main study. The data for the main study was collected at different schools to those of the pilot study. As with the pilot study, the same process of requesting permission was followed; participants, their parents and teachers were informed about the study. The participants were given the opportunity to decide on voluntarily participating in the study, after which the questionnaires were dispensed to groups of learners. As with the pilot study, these groups consisted of 30 to 40 learners at a time, in a class room setting within the limitations of the school time table. Completion of the questionnaires took approximately 45 - 60 minutes.

UNIVERSITY of the WESTERN CAPE

4.11. Data analysis

Analyzing and interpreting data involves drawing conclusions from the data collected and presenting the findings in tables, figures and pictures, to verify the conclusions and provide answers to the researcher's research questions (Creswell, 2009). Descriptive statistics are used to describe or summarize a set of data, while inferential statistics are used to make inferences from the sample chosen to a larger population (Gray, 2009). The key variable in this research study was adolescence decision-making in association with family functioning.

The raw data collected from the research conducted was entered into the Statistical Package for the Social Science (SPSS) V22. The data was coded, cleaned, checked for errors and analysed using descriptive and inferential statistics. Descriptive statistics include frequencies

and means, while inferential statistics include a Pearson correlations and linear regression analysis. Regression analysis is a quantitative research method that is used when the study involves modelling and analysing several variables, where the relationship includes a dependent variable and one or more independent variables to predict effects (McGrath, 2011).

4.12. Reliability and Validity

The test-retest method was implemented in the pilot study, which assisted in measuring the reliability of the instrument. Reliability is an indication of consistency between two measures of the same thing (Black, 1999, as cited in Gray, 2009). For the research tool to be classified as reliable, it is expected to provide the same results when something was measured at different times.

Validity applies to particular interpretations of tests, assessments, questionnaires or other data collection instruments (Arthur et al., 2012). According to David (2010), it is the degree to which whatever is measured is what the researcher intended. To ensure validity, a research instrument must measure whatever it was intended to measure. To achieve validity, the research instrument subject area and the operationally defined subject areas must match exactly (Gray, 2009). The Melbourne Decision-Making Questionnaire has been used in crosscultural research in six different countries. The Cronbach Alpha scores of this cross-cultural research for the various subscales were as follows: vigilance - .80; hyper vigilance - .74; procrastination - .81; and buck-passing - .87 (Mann et al., 1998; Deniz, 2011).

The McMaster Approach to families is a comprehensive model with an integrated set of theoretical constructs, assessment instruments and treatment methods. It has been used

successfully in numerous settings and provides researchers with an empirically validated approach to assessing and treating families (Miller, Ryan, Keitner, Bishop & Epstein, 2000).

4.13. Ethics statement

Permission to conduct the study was requested from the Ethics Committee of the University of the Western Cape. Once ethical clearance was received, permission to collect the data was requested from the Western Cape Department of Education in order to gain access to the various schools. The following ethical guidelines were followed in the pilot study and the main study:

- (i) The purpose of the study was explained to the participants and their parents in the three official languages of the Western Cape being, Afrikaans, English and isiXhosa (see Appendix IV);
- (ii) Written Informed consent was requested from the parents of the participants (see Appendix II) and Assent forms were completed by the participants before participation (see Appendix III),
- (iii) participation in the study was voluntary;
- (iv) participants were informed of their right to withdraw from the study, at any time, without prejudice;
- (v) participants were treated with respect and dignity;
- (vi) all Information sheets, Informed consent and Assent forms, as well as the questionnaires, were available in the three official languages;
- (vii) all questionnaires were coded with numbers to ensure anonymity of participants;

(viii) all information was handled confidentially and only used for the benefit of the

study;

(ix) the questionnaires were collected and kept in a secure place;

(x) there was no known risks involved with participation in the study; and

(xi) the research findings and feedback were disseminated to the various schools that

assisted in the research study.

4.14. Conclusion

This chapter explored the research methodology used to conduct this study. It highlighted the

overall process and procedure followed, in detail, starting at the submission of the research

proposal, to the main study of the data collection and analysis procedure. The results of this

research follow in the next chapter.

UNIVERSITY of the

WESTERN CAPE

CHAPTER 5

RESULTS

5.1. Introduction

Chapter Five presents the results of the analysis of this study. The Statistical Package for the Social Sciences (SPSS) V22 was used to analyse the data that was gathered. The results for this research are provided by means of descriptive and inferential statistics.

5.2. Overview of the analysis

The overview of the analysis lies within the objectives and the hypothesis given below:

5.2.1. Objectives of the Study

The objectives of the study were to:

- Determine the prevalence of the dimensions of the family functioning of adolescents;
- Establish the most prevalent decision-making style of adolescents;
- Examine the role of family functioning in the decision-making styles of adolescents.

5.2.2. Hypothesis

Family functioning has an effect on the decision-making styles of adolescents.

5.3. Inter-correlations

Table: 5.1. Reliability statistics

Variable	Cronbach Alpha Score
Decision-making styles	.60
Family Functioning	.63

Cronbach Alpha coefficients above 0.75 are deemed to be acceptable, while 0.60 is considered to be moderately acceptable (Anastasi, 1982). For this study, with the internal consistency as indicated in Table: 5.1., the Chronbach Alphas are deemed acceptable.

5.4. Demographic Profile

Table: 5.2 is an overview of the demographic information of the 457 participants in the study at the secondary schools within the Overberg area.

UNIVERSITY of the WESTERN CAPE

Table: 5.2: Demographic information of participants

Variable		N = 457	%
Gender	Male	209	46.2
Gender	Female	243	53.8
	Coloured	248	56.4
Race	Black/African	186	42.3
	White	6	1.4
	Afrikaans	253	56.5
Homo longuago	English	3	0.7
Home language	IsiXhosa	166	37.1
	Other	26	5.8
Living with	Two parents	301	69.4
Living with	One parent	133	30.6
	Married	258	57.6
		43	9.6
Marital atatus of paranta	Living together, but not married Single Widowed	72	16.1
Marital status of parents	Divorced	42	9.4
	Other	19	4.2
	Other	14	3.1

The demographic information in Table: 5.2. indicates that there were more female $(n=243 \ [53.8\%])$ than male participants $(n=209 \ [46.2\%])$. The majority of participants identified themselves as Coloured $(n=248 \ [56.4\%])$, followed by Black African $(n=186 \ [2.3\%])$ with the least participants being White $(n=6 \ [1.4\%])$. The majority of participants were Afrikaans speaking $(n=253 \ [56.5\%])$, followed by isiXhosa $(n=166 \ [37.1\%])$ and English $(n=3 \ [0.7\%])$. In terms of family structure, the larger proportion of the participants lived in two-parent households $(n=301 \ [69.4\%])$, and $(n=133 \ [30.6\%])$. In the marital status of parents the majority indicated that their parents were married $(n=258 \ [57.6\%])$ and $(n=43 \ [9.6\%])$ were living together, bur not married. The least of participants $(n=14 \ [3.1\%])$ indicated other.

Table: 5.3: Mean Age of participant

Age	Minimum	Maximum	M	SD
7.90	14	21	16.3 years	1.45

In Table: 5.3. above, the average age of the participants was 16.3 (SD = 1.45) years, of whom the minimum age was 14 and the maximum, 21 years of age.

5.5. Family Functioning

This section presents the Means (M) and Standard Deviations (SD) of the dimensions of family functioning, considered the most important dimensions to determine the functioning of a family (Ryan et al., 2005). Tables: 5.4 to 5.10 provide the Means (M) and Standard Deviations (SD) scores of the six dimensions of the McMaster Model of Family Functioning, and Table: 5.11. provides the total scores. High Mean scores suggest that there are challenges in the family functioning of the participants, while low scores indicate that the family functioning is appropriate. Participants were required to respond that they either agreed or disagreed with the item in the scale. Disagreement was indicated by the highest score of 4.

Table: 5.4. Subscale of General Family Functioning

No.	Items	Means	SD
1	In times of crisis, we can turn to each other for support.	2.15	1.07
6	Individuals are accepted for who they are.	2.36	1.01
11	We can express feelings to each other.	2.31	1.04
16	We feel accepted for who we are.	2.12	1.05
21	We are able to make decisions about how to solve problems.	2.17	1.01
26	We can express feelings to each other.	2.04	1.06
31	Planning family activities is difficult because we misunderstand each other.	2.20	1.03
36	We cannot talk to each other about the sadness we feel.	2.32	1.06
41	We avoid discussing our fears and concerns.	2.33	.97
46	There is lots of bad feeling in the family.	2.25	1.06
51	Making decisions is a problem for our family.	2.26	1.08
56	We confide in each other.	2.06	1.04
Mean	Score:	2.19	.45

Responses were on a scale of: 1 = strongly agree to 4 = strongly disagree

The illustration of scores in Table: 5.4. refer to the general family functioning scores of participants. The highest mean score was indicated for the item; 'We feel accepted for who we are' (M = 2.36, SD = 1.01). This was followed by the items; 'We avoid discussing our fears and concerns' (M = 2.33, SD = 0.97) and 'We can express feelings to each other' (M = 2.31, SD = 1.04). The items with the least responses were; 'We confide in each other' (M = 2.06, SD = 1.04) and 'We can express feelings to each other' (M = 2.04, SD = 1.06).

Table: 5.5. Subscale of Affective Responsiveness

No.	Items	Mean	SD
9	We express tenderness.	2.54	1.05
19	We cry openly.	2.33	1.08
28	We are reluctant to show our affection for each other	2.65	1.01
39	Some of us don't respond emotionally	2.54	1.04
49	We do not show our love for each other	2.18	1.10
57	Tenderness takes second place to other things in our family	2.52	1.02
Mean	Mean Score:		.43

Responses were on a scale of: 1 = strongly agree to 4 = strongly disagree

Table: 5.5. illustrates the scoring of Affective Responsiveness of participants. The majority of participants indicated; 'We are reluctant to show our affection for each other' (M = 2.65, SD = 1.01). This was followed by the perception: 'We express tenderness' (M = 2.54, SD = 1.05); 'Some of us don't respond emotionally' (M = 2.54, SD = 1.04) and 'Tenderness takes second place to other things in our family' (M = 2.52, SD = 1.02). The least responses were for; 'We crying openly' (M = 2.33, SD = 1.08) and 'We do not show love for each other (M = 2.18, SD = 1.10).

Table: 5.6. Subscale of Problem-Solving

No.	Items	Mean	SD
1	We resolve most everyday problems around the house.	2.22	1.00
12	We usually act on our decisions regarding problems.	2.31	.99
24	After our family tries to solve a problem, we usually discuss whether it worked or not.	2.25	1.06
38	We resolve most emotional upsets that come up.	2.34	1.05
50	We confront problems involving feelings.	2.38	1.01
60	We try to think of different ways to solve problems.	2.11	1.07
Mean	Scores:	2.26	.59

Responses were on a scale of: 1 = strongly agree to 4 = strongly disagree

Table: 5.6. presents the Problem-solving dimension of participants. On average, participants stated that in their family; 'We confront problems involving feelings' (M = 2.38, SD = 1.01). This was followed by; 'We resolve most emotional upsets that come up' (M = 2.34, SD = 1.05) and 'We usually act on our decisions regarding problems' (M = 2.31, SD = 0.99). Participants also indicated; 'After our family tries to solve a problem, we usually discuss whether it worked or not' (M = 2.25, SD = 1.06). The least responses on average were: 'We try to think of different ways to solve problems' (M = 2.11, SD = 1.07).

Table: 5.7. Subscale of Communication

No.	Items	Mean	SD
1	When someone is upset the others know why.	2.47	.99
8	People come right out and say things instead of hinting at them.	2.39	1.10
14	We talk to people directly rather than through go-betweens.	2.28	1.04
18	We are frank (direct, straightforward) with each other.	2.17	1.08
22	When we don't like what someone has done, we tell them.	2.17	1.04
29	We sometimes run out of things that we need to say	2.47	.98
35	You can tell what a person is feeling from what they say	2.47	.97
43	It is difficult to talk to each other about tender feelings	2.44	1.05
52	We often don't say what we mean	2.53	.99
59	We don't talk to each other when we are angry	2.44	1.08
Mean	Mean Scores:		.40

Responses were on a scale of: 1 = strongly agree to 4 = strongly disagree

In Table: 5.7., patterns of Communication within the family are indicated. The majority of responses indicated; 'We often don't say what we mean' (M = 2.53, SD = 0.99). This was followed by; 'When someone is upset the others know why' (M = 2.47, SD = 0.99), 'We sometimes run out of things that we need to say' (M = 2.47, SD = 0.98) and 'You can tell what a person is feeling from what they say' (M = 2.47, SD = 0.97). The least responses on average were for; 'We are frank (direct, straightforward) with each other' (M = 2.17, SD = 1.08) and 'When we don't like what someone has done, we tell them' (M = 2.17, SD = 1.04).

Table: 5.8. Subscale of Roles

No.	Items	Mean	SD
1	We make sure members meet their family responsibilities.	2.17	1.03
10	Each of us has particular duties and responsibilities.	2.04	1.01
15	We discuss who are responsible for household jobs.	2.30	1.05
23	We have trouble meeting our financial obligations	2.54	1.01
30	Family tasks don't get spread around enough	2.58	1.04
40	There is little time to explore personal interest		1.01
45	If people asks to do something, they need reminding		1.01
56	We confide in each other	2.52	1.02
53	We're generally dissatisfied with the family duties assigned to us		1.01
58	We don't have reasonable transport		1.07
Mean	Mean Scores:		.39

Responses were on a scale of: 1 = strongly agree to 4 = strongly disagree

The results in Table 5.8. illustrate that the highest responses for the dimension of Roles in the family was for; 'Family tasks don't get spread around enough' (M = 2.58, SD = 1.04) and 'We confide in each other' (M = 2.52, SD = 1.02). The least responses of the participants were for; 'Each of us has particular duties and responsibilities' (M = 2.04, SD = 1.01).

Table: 5.9. Subscale of Affective Involvement

No.	Items	Mean	SD
5	If someone is in trouble, the others become too involved	2.47	1.02
13	We only get the interest of others when something is important to them	2.50	1.01
25	We are too self-centred	2.52	1.00
33	We get involved with each other only when something interests us	2.38	1.08
37	We show interest in each other when we can get something out of it personally	2.56	1.09
42	Our family shows interest in each other when they can get something out of it	2.29	1.10
54	Even though we mean well, we intrude too much into each other's lives	2.40	.99
Mean Score:		2.43	.50

Responses were on a scale of: 1 = strongly agree to 4 = strongly disagree

Table: 5.9. refers to Affective family involvement. The majority of participants indicated that; 'We show interest in each other when we can get something out of it personally' (M = 2.56, SD = 1.09). This was followed by; 'We are too self-centred' (M = 2.52, SD = 1.00) and 'We only get the interest of others when something is important to them' (M = 2.50, SD = 1.01). The least responses on average were for; 'Our family shows interest in each other when they can get something out of it' (M = 2.29, SD = 1.10).

Table: 5.10. Subscale of Behaviour Control

No.	Items	Mean	SD
7	We know what to do in an emergency.	2.02	.98
17	We have rules about hitting people.	2.71	1.07
20	There are rules in our family about dangerous situations.	2.31	1.03
27	We don't know what to do when an emergency comes	2.18	1.02
32	You can easily get away with breaking the rules	2.18	1.04
44	We have no clear expectations about our toilet habits	2.28	1.05
47	We don't hold to any rules or standards	2.33	1.03
48	If the rules are broken, we don't know what to expect	2.55	1.04
55	Anything goes in our family	2.49	1.11
Mean	Score:	2.34	.43

Responses were on a scale of: 1 = strongly agree to 4 = strongly disagree

Table: 5.10. presents the mean scores for the dimension of Behaviour Control. The highest mean score was from participants who indicated; 'We have rules about hitting people' (M = 2.71, SD = 1.07). This was followed by participants who indicated; 'If the rules are broken, we don't know what to expect' (M = 2.55, SD = 1.04). The least responses indicated were from participants who stated; 'We know what to do in an emergency' (M = 2.02, SD = 0.98).

Table: 5.11. Overall functioning of families

Variables	Basic Cut-off value	Mean	SD
Problem Solving	2.2	2.27	.59
Communication	2.2	2.37	.41
Roles	2.3	2.43	.32
Affective Responsiveness	2.2	2.47	.43
Affective Involvement	2.1	2.45	.51
Behaviour Control	1.9	2.35	.47
General Family Functioning (GFF)	2.0	2.27	.39

For the family to be defined as having a problem on any of the scales above, the 'Mean' has to be greater than the established cut-off.

The results in Table: 5.11. illustrate that the families of participants may have ineffective functioning on all the dimensions, as the Means of the variables are greater than the cut-off values. The highest scores, and possibly the most challenging dimensions in the family functioning, was for Affective Responsiveness (M = 2.47, SD = 0.43), Affective Involvement (M = 2.45, SD = 0.51) and Roles (M = 2.43, SD = 0.32) in the family. Although still above the cut-off, the lowest scores were for Problem-solving (M = 2.27, SD = 0.59) and General family functioning (M = 2.27, SD = 0.39) in the family.

5.6. Decision-making styles

This section presents the Means (M) and Standard Deviation (SD) scores of the decision-making styles. Table 5.12 to Table 5.17 provides the means and standard deviation scores of the six decision-making styles and Table 5.18 is the overall scores.

Table: 5.12. Subscale of the Hyper Vigilance Decision-making Style

No.	Items	Mean	SD
1	I feel as if I'm under tremendous time pressure when making decisions.	.85	.64
6	Whenever I get upset by having to make decisions, I choose on the spur of the moment.	1.03	.72
21	The possibility that some small thing might go wrong causes me to swing abruptly in my preferences.		.74
25	I choose on the basis of some small thing.	1.03	.70
31	I can't think straight if I have to make decisions in a hurry.	1.18	.74

The responses were on a scale of: True for me = 2 to Not true for me = 0

Table: 5.12. shows the results of the Hyper Vigilance Decision-making Style. The majority of participants indicated; 'I can't think straight if I have to make decisions in a hurry' (M = 1.18, SD = 0.74). This was followed by; 'The possibility that some small thing might go wrong causes me to swing abruptly in my preferences' (M = 1.03, SD = 0.74) and 'Whenever I get upset by having to make decisions, I choose on the spur of the moment' (M = 1.03, SD = 0.72). The least responses were for; 'I feel as if I'm under tremendous time pressure when making decisions' (M = 0.85, SD = 0.64).

Table: 5.13. Subscale of the Rationalization Decision-Making Style

No.	Items	Mean	SD
2	I feel better about choosing if I can convince myself that the decision is not all that important.	1.03	.78
13	After making a decision I am inclined to undervalue the worth of the alternatives I did not choose.		.70
16	I only want to hear information about my preferred alternative		.74
20	I forget or overlook important information about choice alternatives.	.97	.72
29	After a decision is made I spend a lot of time convincing myself it was correct.	1.50	.66

The responses were on a scale of: True for me = 2 to Not true for me = 0

Table: 5.13. illustrates the scoring on the Rationalization Decision-making Style. The majority of participants indicated; 'After a decision is made I spend a lot of time convincing myself it was correct' (M = 1.50, SD = 0.66) and 'I only want to hear information about my preferred alternative' (M = 1.12, SD = 0.74). The least responses were for; 'After making a decision I am inclined to undervalue the worth of the alternatives I did not choose' (M = 0.87, SD = 0.70).

Table: 5.14. Subscale on the Vigilance Decision-Making Style

No.	Items	Mean	SD
3	I like to consider all of the alternatives.	1.15	.75
7	I try to find out the disadvantages of all alternatives.	1.43	.69
11	I consider how best to carry out the decision.	1.47	.68
14	When making decisions I like to collect lots of information.	1.55	.66
19	I try to be clear about my objectives before choosing.	1.57	.63
24	I take a lot of care before choosing.	1.40	.67

The responses were on a scale of: True for me = 2 to Not true for me = 0

The results in Table: 5.14. refer to the Vigilance Decision-making Style. Most participants chose; 'I try to be clear about my objectives before choosing' (M = 1.57, SD = 0.63) and 'When making decisions I like to collect lots of information' (M = 1.55, SD = 0.66). The least responses were; 'I like to consider all of the alternatives' (M = 1.15, SD = 0.75).

Table: 5.15. Subscale on the Defensive/Avoidance Decision-Making Style

No.	Items WESTERN CAPE	Mean	SD
4	When I have a decision to make I try not to think about it.	.77	.76
10	I feel uncomfortable about making decisions.	1.01	.74
15	I avoid making decisions.	.68	.71
23	Whenever I face a difficult decision, I feel pessimistic about finding a good solution.	1.21	.68
26	I don't make decisions unless I really have to.	1.07	.81

The responses were on a scale of: True for me = 2 to Not true for me = 0

Table: 5.15. presents the Defensive/Avoidance Decision-making Style. The majority of participants indicated; 'Whenever I face a difficult decision, I feel pessimistic about finding a good solution' (M = 1.21, SD = 0.68). The least amount of participants chose; 'When I have a decision to make I try not to think about it' (M = 0.77, SD = 0.76) and 'I avoid making decisions' (M = 0.68, SD = 0.71).

Table: 5.16. Subscale on the Buck-Passing Decision-Making Style

No.	Items	Mean	SD
5	I prefer to leave decisions to others.	.63	.75
8	I am inclined to blame others when decisions turn out badly.	.67	.73
18	I don't like to take responsibility for making decisions.	.79	.78
22	If a decision can be made by me or another person I let the other person make it.	.70	.78
28	I prefer that people who are better informed decide for me.	.80	.79

The responses were on a scale of: True for me = 2 to Not true for me = 0

Table: 5.16. provides the Buck-passing Decision-making Style. The majority participants indicated; 'I prefer that people who are better informed, decide for me' (M=0.80, SD=0.79) and 'I don't like to take responsibility for making decisions' (M=0.79, SD=0.78). The least participants chose; 'I prefer to leave decisions to others' (M=0.63, SD=0.75).

Table: 5.17. Subscale on the Procrastination Decision Making Style

	UNIVERSITY of the					
No.	Items WESTERN CAPE	Mean	SD			
9	I waste a lot of time on trivial matters before getting to the final decision.	1.04	.69			
12	Even after I have made a decision I delay acting upon it.		.75			
17	When I have to make a decision I wait for a long time before starting to think about it.	1.05	.79			
27	I delay making decisions until it is too late.	.78	.76			
30	I put off making decisions.	.79	.71			

The responses were on a scale of: True for me = 2 to Not true for me =

The illustration of scores in Table: 5.17. refers to the Procrastination Decision-making Style. The highest score of participants indicated; 'Even after I have made a decision I delay acting upon it' (M = 1.11, SD = 0.75) and 'When I have to make a decision I wait for a long time before starting to think about it' (M = 1.05, SD = 0.79). The least amount of participants indicated item; 'I delay making decisions until it is too late' (M=0.78, SD=.76).

Table: 5.18. General Overview of the Decision Making Styles

Items	Mean	SD
Vigilance	.71	.17
Hyper vigilance	.51	.18
Defective Avoidant	.47	.19
Buck-passing	.36	.22
Procastination	.48	.19
Rational	.55	.17

The results in Table: 5.18. reflect *Vigilance* as the most prevalent decision-making style (M = 0.71, SD = 0.17), and *Buck-passing* as the lowest (M = 0.36, N = 0.22).

5.7. Relationships between the variables

This section provides the results of the relationships between the variables. A correlation (Table: 5.19.) was first conducted to determine the relationship between the dimensions of family functioning and the different decision-making styles. This correlation was then followed by separate regression analyses (Table: 5.20.) to determine whether family functioning predicted the decision-making style of adolescents.

	Variables	_	8	က	4	5	ဖ	7	ω	6	10	1	12
-	Problem Solving	-											
2	Communication	.13*											
က	Roles	*81	.17										
4	Affective Responsive	*.16	04	.16*									
2	Affective Involvement	28	23	.23**	UŅI WES								
9	Behaviour Control	*.15	60:-	80.	VĢ.R	.22**							
7	General Family Func.	05	.05	.19*	60. N	70.	07						
8	Vigilance	80.	03	.04	02	05	00.	90:					
6	Hyper vigilance	90.	10	.02	90:-	80:	90:	.02	.05				
10	Defensive Avoidance	90:-	05	04	10.	70.	.10	10	12**	.22**			
11	Buck Passing	01	.03	13	10.	80.	03	*11.	15	.24**	.37**		
12	Procrastination	90.	02	60'-	05	02	05	03	03	.28*	.22**	.29**	
13	Rationalization	60.	60'-	04	80:-	07	04	04	.10*	.26**	** 41.	.13*	.25**
(

*. Correlation is significant at the 0.05 level (2-tailed) **. Correlation is significant at the 0.01 level (2-tailed)

The results in Table: 5.19. show that there are no significant positive relationships between the dimensions of family functioning and the decision-making styles of adolescents. There were significant negative relationships to be found between Communication and Hyper vigilance $(r = -.10^*)$, Roles and Buck-passing $(r = -.13^*)$, General family functioning and Defensive avoidance $(r = -.10^*)$ and Buck-passing $(r = -.11^*)$.

In Table: 5.20. the regression analyses suggest that family functioning has an effect on only Defensive avoidance. Specifically, Problem-solving ($\beta = 2.27$, p = .02) and Communication ($\beta = 2.54$, p = .01) were significantly positive predictors of a Defensive avoidance decision-making style. The final model accounted for 4% of the variance of a Defensive avoidance decision-making style.

Table: 5.20. Regression analyses predicting decision-making styles of adolescents

Family Functioning	b	SE b	β	t	p
	WES Vig	ilance	E		
Constant	.88				
Problem-Solving	01	.02	03	45	.65
Communication	02	.03	05	84	.40
Roles	03	.03	06	-1.04	.30
Affective Responsiveness	.03	.03	09	-1.49	.14
Affective Involvement	.03	.02	08	1.34	.18
Behavioural Control	03	.02	07	-1.14	.25
General Family Functioning	.03	.03	.07	.99	.32
Hyper vigilance					
Constant	Constant .45				
Problem-Solving	.03	.02	.08	1.18	.24
Communication	.03	.03	.06	.96	.34
Roles	02	.03	04	70	.49
Affective Responsiveness	.03	.03	.06	1.04	.30
Affective Involvement	00	.02	01	17	.86
Behavioural Control	05	.02	12	-1.89	.06
General Family Functioning	.02	.03	.04	.64	.53

	Buck-	passing			
Constant	.14	F 3			
Problem-Solving	.03	.03	.07	1.05	.29
Communication	.03	.03	.07	1.02	.31
Roles	.01	.04	.02	.31	.76
Affective Responsiveness	03	.03	05	85	.40
Affective Involvement	00	.03	01	13	.90
Behavioural Control	.05	.03	.12	1.92	.06
General Family Functioning	01	.04	01	18	.85
		Avoidanc			
Constant	.33	Avoidano			
		.02	.15	2.27*	02
Problem-Solving Communication	.05	.02	.15	2.21*	.02
	04	.03	_	-1.11	.01
Roles	.02	.03	07 .05	.86	.39
Affective Responsiveness Affective Involvement	02	.03	04	70	.49
			3		
Behavioural Control	00	.03	01	09	.93
General Family Functioning	03	.03	06	84	.40
Procrastination					
Constant	43 UN-43	SITY of th	e		
Problem-Solving		.02	.02	.21	.83
Communication	03	.03	05	82	.41
Roles	.01	.04	.01	.20	.85
Affective Responsiveness	00	.03	00	04	.97
Affective Involvement	.00	.03	.00	.07	.94
Behavioural Control	.01	.03	.02	.27	.79
General Family Functioning	.02	.03	.05	.69	.49
	Ration	alization			
Constant	.39				
Problem-Solving	.00	.02	.00	.03	.98
Communication	.04	.03	.09	1.37	.17
Roles	03	.03	06	-1.01	.31
Affective Responsiveness	.03	.03	.08	1.27	.20
Affective Involvement	.02	.02	.05	.87	.38
Behavioural Control	01	.02	04	57	.57
General Family Functioning	.03	.03	.06	.84	.40

Note Vigilance: $\Delta R^2 = 0.01$

Note Hyper vigilance: $\Delta R^2 = 0.01$

Note Buck passing: $\Delta R^2 = 0.01$

Note Defensive Avoidance: $\Delta R^2 = 0.04$

Note Procrastination: $\Delta R^2 = -0.02$

Note Rationalisation: $\Delta R^2 = 0.00$

5.8. Conclusion

The results on the relationship that existed between variables forming part of this research

indicate no positive relationship between the dimensions of family functioning and the

decision-making styles of adolescents. There is however significant negative relationships

found between Communication and Hyper vigilance decision making style. The Hyper

vigilance style refers to the application of an optimistic views towards the various alternatives

of a decision needed to be made, panic and stress sets in and an objectionable alternative is

considered. Another significant negative relationship was found between Roles and Buck-

passing decision making style. The buck-passing decision-making style refers to the passing

of a decision on to someone else. The following chapter includes the discussion, conclusion

and recommendations based on the finding in this chapter.

88

CHAPTER 6

DISCUSSION, CONCLUSION AND RECOMMENDATIONS

6.1. Introduction

The aim of this study was to examine the association between the role of family functioning in the decision-making styles of adolescents in the Overberg Area. This chapter discusses the findings of this research project, in line with the conceptual framework underpinning the study.

6.2. Family Functioning and adolescence

The first objective of this study was to determine the most prevalent dimension of family functioning of adolescents. The results are only based on the adolescents' perception of their families' functioning. In addition, this study sought to determine how well families were functioning. The results of this study indicated that the families of participants might have had ineffective functioning in all the dimensions of the McMaster Model of Family Functioning (MMFF). The dimensions of this Family Assessment Device (FAD), which include problem-solving, communication, roles, affective responsiveness, affective involvement and behaviour control, are considered to have the most important impact on the family (Walsh, 2003). These dimensions are further elaborated upon as follows:

(a) **Problem-solving** includes the family's ability to resolve problems to a level that maintains effective family functioning (Aarons, McDonald, Connelly & Newton, 2007). Effective families solve most problems rapidly, easily and without much thought, so that at times there could be some difficulty eliciting and detailing the

problem-solving steps they follow (Sheriif et al., 2010; Epstein et al., 1983; Walsh, 2003). The stages in this process of problem-solving are as follows:

- (i) Identification of the problem;
- (ii) Communication of the problem to appropriate resources within or outside of the family;
- (iii) Development of alternative action plans;
- (iv) Decision regarding a suitable action;
- (v) Action;
- (vi) Monitoring that action which is taken; and
- (vii) Evaluation of the success of the action.
- (b) **Communication** is defined as the way in which the family exchanges information, and includes consideration for such variables as content, the potential for multiple messages, checking that the communication sent was appropriately conveyed and interpreted by the receiver.
- (c) **Roles**, also referred to as Role allocation or Family roles, are the repetitive patterns of behaviour that underpin the individuals' fulfilment in family functioning. The more functions they adequately fulfil and clearly allocate house accountability mechanisms, the healthier the family.
- (d) **Affective responsiveness** is defined as the ability to respond to a range of stimuli with the appropriate quality and quantity of feelings. The more effective the families' affective responsiveness, the wider the range and more appropriate their responses in terms of quantity and quality for a given situation.
- (e) **Affective involvement** is defined as the degree to which the family shows interest in, and values the activities and interests of, family members.

(f) **Behaviour control** refers to the pattern the family adopts for dealing with the behaviour of individuals within the family. These patterns of behaviour can be seen in three specific situations: (i) physically dangerous situations, (ii) situations involving the meeting and expressing of psychobiological needs and drives, and (iii) situations involving socializing behaviour both inside and outside the family.

The responses of the FAD are coded 1-4 for each subscale, summed up for positively worded items and divided by the total number of items in that scale – the negative worded items are reversed. Individual scale scores range from 1.0 for best functioning and 4.0 for the worse functioning (Miller, Ryan, Keitner, Bishop, & Epstein, 2000). The cut off score indicated in Table: 6.1. for Affective Responsiveness is 2.20, Affective Involvement is 2.10 and Roles is 2.30. For each of the six scales (dimensions of family functioning), Table: 6.1. presents the cut-off value of the McMaster Family Functioning Assessment Device and one added General Family Functioning scale.

Table: 6.1. McMaster Model of Family Functioning & General Functioning Assessment

Scale	McMaster cut-off values	Mean for this study
Problem Solving	2.2	2.27
Communication	2.2	2.37
Roles	2.3	2.43
Affective Involvement	2.1	2.47
Affective Responsiveness	2.2	2.45
Behaviour Control	1.9	2.35
General Functioning	2.0	2.27

The results of this study indicated that the highest scores, and possibly the most challenging dimensions in the family functioning of participants, were Affective Involvement, Affective Responsiveness and Roles. Affective Involvement refers to how often and to what level

family members show an interest and invest themselves in each other (Aarons et al., 2007), through particular activities of individuals (Ryan et al., 2005). The focus of this dimension rests on the degree to which, and the manner in which, the members show interest. This can range from a total lack, to an extreme amount, of involvement. There are several styles of Affective Involvement, of which Empathic Involvement (members demonstrate a true affective concern for the interests of others in the family, even though those concerns may be peripheral to their own interests) is the most healthy form of functioning: without it, the family becomes less effective in their functioning (Ryan et al., 2005). Roche, Ensminger & Cherlin, (2007) conducted a study to determine the variations of parenting and adolescent outcomes of families living in low income urban areas. The results indicated that higher levels of parental involvement were associated with lower levels of psychological problems, and ineffective involvement contributed to adolescents' problem behaviour and lack of school performance. In a systematic review to examine the link between anti-social behaviour of adolescents and parental practice, Human-Hendricks & Roman (2014) found that consistent parental involvement, attachment, role allocation and affection contribute to a reduced negative behaviour.

Affective Responsiveness is also defined as the families' ability to respond to a full range of experiences, in an appropriate quantity and quality of feeling (Aarons et al., 2007). According to Ryan et al. (2005), this level of Affective Responsiveness is determined by observing the experience of affective responses to effective stimuli: whether or not family members are able to respond with a full spectrum of feelings experienced or the emotion experienced is consistent or appropriate with the stimulus or situational context.

Roles refer to the repetitive patterns of behaviour through which family functions are fulfilled by family members (Aarons et al., 2007). Ryan et al. (2005) identify five necessary functioning in this dimension:

- (1) provision of resources (i.e. food, clothing and shelter);
- (2) nurturing and support (i.e. comfort and warmth);
- (3) adult sexual gratification;
- (4) personal development (i.e. assisting with school work, finding a career and guidance);
- (5) maintenance and management (i.e. decision making functions, boundary and membership functions, behaviour control functions, household finance functions and health related functions).

In addition, two further integral issues of role functioning include:

- (1) role allocation referring to the family's pattern in assigning roles; and
- (2) role accountability which refers to the way in which the family ensures that these functions are done.

In another study that assessed the family functioning of Caucasian and Hispanic American youth, Aarons et al. (2007) found that Hispanic families scored significantly higher on the Roles and Affective Responsiveness domains. The responses in the areas of inter and intrapersonal strength, school functioning and affective strength for Hispanic youth was significantly higher as well. A study by McCreary & Dancy (2004) on the dimensions of family functioning concluded that family interactions that consists of Affective Involvement, Affective Responsiveness, Roles and Communication, demonstrate effective family functioning. Whereas ineffective functioning consists of families that demonstrate uncaring,

hostility and violence, lack of communication, lack of affective involvement and affective responsiveness, as well as a lack of roles.

In contrast, the lowest scores, still above the cut-off value, however, were for Problem-solving and General Family Functioning. Problem-solving is referred to as a family's sufficient ability to resolve problems to a level that maintain effective family functioning (Aarons et al., 2007; Ryan et al., 2005). Ryan et al. (2005) further elaborate that family problems are issues that threaten the family's functional capacity and which the family may have difficulty in solving. Family problems can be divided into instrumental problems that refer to everyday life problems (managing finances, obtaining food, etc.) and affective problems, which refer to issues of emotions (feelings) such as anger, depression, etc.

General family functioning refers to the overall level of family health/pathology and functionality (Petrocelli, Calhoun & Glaser, 2003). This dimension was a product of the Family Assessment Device (FAD) (Ryan et al., 2005), a subscale separate from the other six subscales (Petrocelli et al., 2003) most effectively used as a research tool (Ryan et al., 2005). According to Petrocelli et al. (2003), a family that displays high levels of general family functioning is known to be healthy, lack psychosocial problems, develop cohesion, can adapt well to challenges and sets clear roles and boundaries. The study of Petrocelli et al. (2003) aimed to determine the role of general family functioning in the quality of family relationships. The results of their study indicated that the Affective Involvement subscale was not significantly correlated to the quality of family relationships. The results further suggested that families may be able to problem solve, set clear behavioural roles, respond with appropriate affection, and maintain desired standards. If they cannot maintain these, however, they may then perceive their overall relationship as problematic.

According to Miller et al. (2000), these dimensions of family functioning were formulated to create an understanding of the structure, organization and transactional patterns associated with family difficulties. Not one of these dimensions, however, are considered as the foundation for conceptualizing family behaviour: they merely contribute to create an overall understanding of the family in which some areas of functioning may overlap or interact with one another (Ryan et al., 2005).

Unhealthy family functioning is based on the Mean and Standard Deviation being greater than or equal to the cut off-scores in that dimension, and less than the cut-off score indicates healthy family functioning (Ryan et al., 2005). The scoring in this study is above the cut-off value, which implies possible ineffective functioning of the family in these dimensions. It could, therefore, be concluded that the overall family functioning of the participants was ineffective in one or more of these dimensions.

The FAD was designed as a screening instrument to identify problem areas in the simplest way. According to Epstein, Baldwin & Bishop (1983), these dimensions of family functioning are not dependent on one another. Challenges experienced in one dimension of family functioning, may not have an impact on another, i.e. some families may have challenges with behaviour control, another with communication, and some may even have challenges with both these dimensions. The implications of the results of this research suggests that ineffective family functioning in any of these dimensions, according to Ryan et al. (2005), can become a contributing factor to the emotional and physical problems of family members. Effective functioning in all of these dimensions on the other hand supports the optimal physical and emotional well-being of family members.

According to Ryan et al. (2005), family functioning is dynamic and generally refers to the quality of family life concerned with the wellness, competence, strengths and weaknesses of family members (Shek, 2002). Most importantly, family functioning is also considered to have an impact on the developing adolescent (Koen, Van Eden & Venter, 2011). Family factors such as divorce, single parent families, family disintegration, lack of involvement, poor communication can influence adolescent's attitudes and behaviour (Louw, Van Ede & Louw, 1998). Also considering that the family is seen as a working unit that is greater than the sum of its parts, each individual affects the family as a whole, while the family members also affect each individual. Good family relationships are important protective factors: adequate emotional sharing; high flexibility in rules; good levels of satisfaction of all family members; all contribute to adolescent's well-being (Tafa & Baiocco, 2009).

The study of Shek (2002) on the association between family functioning and adolescent adjustment revealed that family functioning was significantly related to adolescents' psychological well-being, which included existential well-being, life satisfaction, self-esteem, and a sense of mastery. The results further indicated that adolescents, who perceived the family as dysfunctional, displayed more problem behaviour in both poor and non-poor groups. Henderson, Dakof, Schwartz & Liddle (2006) investigated direct and indirect effects of family functioning on the severity of adolescent externalizing problems. Their findings indicated that family functioning exerted direct and shared effects on externalizing problems.

Similarly, Parra, Oliva & Reina (2013) conducted a longitudinal study with a group of adolescents. In this study the relationship between adolescents and their parents were analysed over a period of 10 years. The results indicated that the levels of communication and affection diminished, however, their perceived adaptability remained and cohesion increased

as the frequency of conflict increased. Continuous and unresolved conflict may result in serious self-esteem, autonomy and self-control deficiencies in adolescence (Koen, van Eden, & Venter, 2011).

Based on a South African study by Holborn & Eddy (2011), many South African children are growing up in dysfunctional families because of the unique conditions that affected the social structure: social stressors and historical factors; the social, political and economic conditions of colonialism; and the urbanisation of South African families. These conditions have led to changes in the structure, responsibilities and roles of the families (Makiwane et al., 2012). They have also created conditions of weakened family ties, diminished obligations towards one another, as well as a lack of safety and security which further exacerbates the family dysfunctionality. According to Amoateng et al. (2004), a gap is formed between idealization of the family and reality, as often children's living conditions affect their impressions of what families are and should be. Holborn & Eddy (2011) further asserts that dysfunctional families could damage the adolescent's desire to prosper.

Fostering adolescents' self-endorsed functioning through empathy, giving choices wherever possible, and encouraging them to act upon their personal values and interests, seems to be crucial in dealing successfully with the challenges of raising adolescents (Van Pentegem et al., 2012). Families need to allow adolescents to make choices and not dominate decision-making (Carpenter & Mulligan, 2009). It may foster adolescents' individualisation from their parents, to try more things on their own and develop competence and efficacy (Eccless & Gootman, 2002).

6.3. Decision-making styles of adolescence

The second objective for this study was to determine the most prevalent decision-making style of adolescents. According to Janis & Mann (1977, in Mann et al., 1977), decision-making styles, also referred to as coping patterns, are strategies that individuals use to deal with threats and stress, or, coping behaviour when challenged with decisions that have to be made. The various decision-making styles are: Vigilance, Hyper vigilance, Defective avoidant, Buck-passing and Procrastination. Decision-making styles are also defined as, a situation encountered, which includes the approach, reaction and action of an individual who is about to make a decision (Phillips, Pazienza & Ferrin, 1984, as cited in Deniz, 2011). Janis & Mann (1977, in Mann et al., 1977) defines these styles as follows:

- (i) The Vigilance decision-making style is applied when an individual is optimistic about finding an alternative solution or answer to a conflicting decision that needs to be made and believes that there is time to follow the steps in making a good choice.
- (ii) The Hyper vigilance style is applied when the individual is optimistic about the various alternatives to the decision that needs to be made, but believes there is insufficient time to search for a possible alternative or solution, so panic and stress sets in and an objectionable alternative is considered.
- (iii) The Defensive avoidance style is applied when an individual is pessimistic about the alternative or solution and adopts either (a) The procrastination decision-making style, delaying or postponing the decision-making, or (b) The buck-passing decision-making style, passing the decision-making responsibility on to someone else (Burnett, 1991).

The results of the current study indicated that the Vigilance decision-making style was the most prevalent among adolescents, while the Buck-passing decision-making style was the least prevalent. The Vigilance decision-making style, is identified as one that encompasses high cognitive activity and is also regarded as a problem-solving strategy (Mann et al., 1997). Buck-passing is one of the Defensive avoidance decision-making styles (Mann et al., 1997) and refers to someone who avoids the responsibility of making a decision by passing it on to someone else (Brown, Abdallah & Ng, 2011).

According to the Janis & Mann (1977, in Mann et al., 1977) model of decision-making styles, effective decision makers use a vigilant pre-decisional strategy, while maladaptive strategies are characterised by procrastination and avoidance. Berzonsky & Ferrari (1996) investigated the relationship between identity orientation and decisional strategies of adolescents. They define identity orientation as the manner in which individuals approach (information orientated) or avoid (avoidant orientation) a task associated with developing their selfidentity. The results indicated that those who reportedly take on an informational orientated approached to self-construction and identity issues used a more cognitive, vigilant decisional strategy, whereas an avoidant orientation approach resorts to maladaptive decision-making strategies, such as avoidance. The vigilant decision-making style approach indicates that information orientated individuals rely on a more deliberate, committed approach, by actively seeking out, evaluating, and utilizing relevant information to deal with decisional conflicts. A study by Luyckx, Soenens, Berzonsky, Smits, Goossens & Vansteenkiste (2007) examined the moderating role of autonomy, self-reflection and self-remuneration. Their aim was to do this in comparison with identity orientated identity processing, identity consolidation and individual well-being. Their findings revealed that individuals, who were more autonomous

in their actions, increased their consolidated identity as a result of an automatic increase in their exploratory actions and improved sense of self-worth.

It is important to bear in mind that there is not much literature available on South African adolescents' decision-making styles. Previous studies on decision-making mainly focus on career decision-making (Martinez, 2007), consumerism (Kamaruddin & Mokhlis, 2003), academic related decisions (Miller & Byrnes, 2001), and other life satisfaction related decisions (Cenkseven-Önder, 2012). These studies focused on the process that adolescents follow to make decisions and also how, having good or bad decision-making abilities, impacted on them. To motivate the importance of acquiring a process of making effective decisions, literature obtained emphasizes the development of decision-making skills as being the ultimate goal during adolescence (Miller & Byrnes, 2001; Eccles & Gootman, 2002). According to Deniz (2011), it is important that adolescents acquire the skills to make confident decisions, as they are in the developmental phase where critical decision-making on the most important key life transitions occur. Full transition through this stage results in the ability to: draw conclusions from available information and predict possible outcomes; reason theoretically; and understand more abstract concepts, such as love and values (Fantasia, 2008) and help guide the exposure of adolescents to experience either treacherous places or exciting adventures (Siegel & Hartzel, 2014).

Deniz (2006) aimed to determine the relationship between decision-making styles and life satisfaction. The findings indicated that there was no significant correlation between life satisfaction, decisions, self-esteem and the avoidance decision-making styles. There was a positive relationship found between problem-focused coping and the vigilance decision-making style, and a negative relationship between problem-focused and buck-passing.

According to Dewberry, Juanchich & Narendran (2013), decision-making competence in everyday life is associated with decision-making styles. Their study aimed to examine decision-making competence in everyday life. The results indicated that cognitive styles do not seem to offer an increase in the validity, while personality does offer an increase in the validity, of decision-making styles for predicting decision-making competencies. These decisions include choices on education; occupation; health; starting a family; and participating in society (Makiwane & Kwizera, 2008). These decisions have a long-term impact on many of life's contexts for adolescence (Seiffge-Krenke, 2002), and how these decisions are made, holds definite implications across all facets of our lives (Brown, Abdallah & Ng, 2011). Therefore having the ability to make wise, intellectual and mature decisions is considered fundamental to living a prosperous and accomplished life (Armesh, 2013).

Decision-making capacities of adolescents are limited, according to Kambam & Thompson (2009), and have been common knowledge for centuries. Typical adolescents have deficiencies in their decision-making capabilities, as a result of a deficiency in their cognitive and psychosocial characteristics, related to their level of development. Adolescents are particularly susceptible to the potentially deleterious effects of emotions on decision-making, as they often make decisions under the influence of emotions and social aspects (Lizarraga, Baquendano, Oliver, & Closas 2009). Decision-making in adolescence may, however, be influenced by other variables as well (Kambam & Thompson, 2009). According to Lizarraga et al. (2009), these variables can be characterized by at least three sources: those associated with the decision; those associated with the decision maker; and those associated with the environments in which the decision occurs. Brown, Abdallah & Ng (2011) identify these variables as the family, friends, and social institutions. The family, according to Zuberi

(2013), however, provide a more immediate and essential context of development for the adolescent. Thus, it can be concluded that the effectiveness and management of decision-making capabilities may have been taught at home (Roman & Davids, 2013).

6.4. Relating family functioning to decision-making styles during adolescence

The third and final objective in this study was to examine the role of family functioning on decision-making styles of adolescents. According to the Department of Social Development (2013), the theoretical underpinnings that draw the connection between family functioning and decision-making, are derived mainly from the general family literature, such as The White Paper on Families. While the family is considered central to the socialization of children (Department of Social Development, 2013), not much is known about the role of family functioning in the decision-making styles of adolescents.

The results of this study suggest that family functioning only has an effect on the Defensive Avoidance decision-making style. The Defensive Avoidance style is considered when an individual is pessimistic about the alternative or solution to a decision and adopts the: (a) Procrastination decision-making style, where the individual delays or postpone making the decision; or the (b) Buck-passing decision-making style, where the responsibility is passed on to someone else (Burnett, 1991). The problem-solving dimension (defined as the ability to resolve problems at a level that maintains effective family functioning) and communication (the manner in which information is exchanged within and between the family members) (Walsh, 2003), were on the other hand significant positive predictors of the defensive avoidance decision-making style. In relation to the McMaster Model of Family Functioning, 4% of the variance of a Defensive Avoidance decision-making style was accounted for. Cenkseven-Önder (2012) examined the influence of decision-making styles on adolescents

and life satisfaction. His findings suggest that the Vigilance and Complacency decisionmaking styles were positively related to family satisfaction.

The results of this current study show that there are no significant positive relationships between the dimensions of family functioning and decision-making styles of adolescents. There are significant negative relationships found between Communication and Hyper vigilance; Roles and Buck-passing; General family functioning and Defensive Avoidance (buck-passing). The regression analyses suggest that family functioning only has an effect on Defensive Avoidance. Specifically, problem-solving and communication were significantly positive predictors of a Defensive Avoidance decision-making style. This may imply that family functioning may not have an impact on the decision-making style of the adolescent, but rather on the process in making affective decisions.

According to Berzonsky & Ferrari (1996), the Defensive Avoidance style was significant to those who experienced pre-decisional panic, low cognitive competence, procrastination and maladaptive decisional strategies. In essence, it further implies that growing up in a troubled family environment, depriving adolescents of good role models (parents), or disturbing the learning patterns (lack of care), are family circumstances that contribute to the development of poor decision-making competence. According to Isen (2001), positive affect enhances decision-making, leading to the development of more flexible, innovative, creative and efficient cognitive processing ability, as long as the task at hand is meaningful or important to the decision maker.

According to literature, the family faces a variety of transitions that demand adaptive efforts by its members, of which adolescence will be the most important transition (Parra, Oliva, &

Reina, 2013). Based on the physical, cognitive and emotional changes that adolescents experience, they can, and do, open the family to a whole new array of values (Cox & Paley, 1997). Characterised by a process of co-evolution, parents and adolescents should engage in these developmental challenges together (Tafa & Baiocco, 2009). It can be challenging for the adolescent, as well as the family members who render the support (Siegel & Hartzel, 2014). The role of the family therefore seems crucial in the development of adolescents (Koen, Van Eden & Venter, 2011).

Parker & Fischhoff (2005) conducted a study that examined to what extent individuals showed consistent performance differences across typical behavioural decision-making tasks, and how those correlated with real world-decision-making. Their findings indicated that higher decision-making competence scores were associated with a positive intact family environment, more constructive cognitive styles and less maladaptive risk behaviour.

UNIVERSITY of the

Amesh (2013) states that the family environment and the relationships that individuals have with family members influence decision-making. Families that emphasise personal decision-making, inspire educational, occupational, recreational, marital and family choices (Zarret & Eccles, 2006). Decisions about aspects of youth and family life are considered important indicators of the family process and contribute to later development. Families need to allow more decision-making and not dominate decision-making (Carpenter & Mulligan, 2009).

6.5. The implications of these results for the adolescents

As mentioned in chapter 2, adolescence is seen as the longest developmental stage, beginning with puberty and extending into the early 20's (Fantasia, 2008); and is largely characterised as beginning and ending the teenage stage (Department of Social Development, 2013).

Adolescence is a confusing period of development (Siegel & Hartzel, 2014) that encompasses a host of new challenges (Bandura, 2005). This phase is most commonly known as a period of storm, stress and experimentation for adolescents (Roman & Frantz, 2013), who have to manage major biological, educational and social roles simultaneously (Bandura, 2005). They have to learn how to deal with puberty, physical changes, emotional invested relationships, the emergence of sexuality, educational transitions, etc. As they mature they are obliged to assume more responsibility and decision-making. The outcome of these changes and the management of these challenges depend on how adolescents develop and exercise their personal power of producing an effect (Bandura, 2005).

According to Siegel & Hartzel (2014), adolescents' experiences are primarily dependant on cognitive changes in their development. These cognitive changes influence their minds and relationships in the way they remember, think, reason, focus attention, make decisions and relate to others (Siegel & Hartzel, 2014). They need to commit themselves to goals that give them a sense of purpose and meaning in life (Roman & Davids, 2013). Goal setting is widely viewed as an important component of the decision-making process (Golatti, Giner, AltenBaumes, Geertz, Ropp & Woulfe, 2006). It also helps them to organise their lives, provide a sense of meaning, not only in their lives, but also in their activities, creating a sense of tolerance in achievement (Bandura, 2005). Eccles & Gootman (2002) add that the most important cognitive changes during this period relates to their ability to think theoretically: consider the hypothetical, as well as the real; process information in a complex and more detailed manner; consider the multiple dimensions of a problem; and have the ability to reflect on the self and on complicated situations.

One of the key tasks associated with this developmental phase is the achievement of a sense of autonomy, also referred to as the ability to make decisions (Van Pentegem, Beyers, Vansteenkiste & Soenens, 2012). Miller & Byrnes (2001) also incorporates their need for independence as they are required to make more decisions on their own. The ultimate goal of promoting independence is to encourage adolescents to make their own decisions and simultaneously enable their behavioural, emotional and cognitive independence (Soenens & Vansteenkiste, 2009). According to Luyckx, Soenens, Berzonsky, Smits, Goossens & Vansteenkiste (2007), the adoption of an information-orientated approach to decision-making is known to be beneficial to adolescents. The avoidant orientated individuals are prone to be reluctant and defensive when dealing with their decisional conflicts. When procrastination persists, the situational demands and consequences eventually determine behavioural actions.

In order to make decisions, a sense of independence that can be acquired through a developmental process, is needed; which is exactly what happens during adolescence. This transformation is achieved by acquiring decision-making skills, including individuality and growth, in association with adult responsibility (Roman, Human & Hiss, 2012).

Ultimately, adolescence is defined as a stage of life when most biological, cognitive, psychological and social characteristics are transformed from a child's to that of an adult (Ogwo, 2013). Adolescents are exposed to these developmental balances, experiences and the added difficulties of the adult world, both as individuals and as a group (Ogwo, 2013). As adolescents navigate and experience their social world, they begin to gain a better understanding of their own behaviour and tendencies, and can therefore become skilled at more accurately predicting or presumably controlling their decision-making behaviour (Albert & Steinberg, 2011). Decision-making behaviour is related to motivational processes,

such as stress-coping patterns; cognitive styles, which include systematic versus non-systematic, or open versus close mindedness, and personality dimensions such as dependence, optimism/pessimism and efficacy (Mann et al., 1997). These cognitive, social and emotional developments require them to: make their own choices, develop their own opinions, foster some form of independence and responsibility, and also, demand less care and attention from the family (Human-Hendrick & Roman, 2014). Furthermore, children need a supportive family environment, throughout their development, that enables them to acquire the problem-solving skills and competence for a well-adjusted life (Roman, 2014).

6.6. Theoretical implications

Family Systems Theory implies that family functioning is related to the accomplishment of essential functions and tasks. These functions and tasks are referred to as the dimensions of family functioning (Walsh, 2003). It further reflects the quality and satisfaction with relations, rather than problem-solving. This approach acts as a screening device for families who may not recognize risks at an early stage (Akister & Stevenson-Hinde, 1991). The Family Assessment Device was formulated to assess family functioning based on all the dimensions of the MMFF and includes a General Functioning Scale that measures the overall level of family functioning. This assessment is based on the individual members' perception of their families functioning. These items are purposely scrambled and match only one dimension which may identify healthy or unhealthy functioning (Ryan et al., 2005). It further aims to reflect the difficulties that a family may be experiencing in the family system. These difficulties can be determined in the dimensions of family functioning.

The family system is a well-established model, commonly used to assess the family systems framework (Buswell et al., 2012). It is commonly examined through the family systems theoretical perspective which includes:

- (1) Family relationships, according to the systems framework, function independently (Morgaine, 2001) and are governed by boundaries or explicit rules for accessing resources and support within the family (Sturge-Apple et al., 2010). There are predictable patterns of interaction within the family system that assist in maintaining the family's equilibrium (Morgaine, 2001). Amoateng et al. (2004), state that the quality and level of family relationships have a direct effect on the behaviour, social competence and goal directedness of individuals. According to Bandura (2005), a family's functioning consists of interlocking relationships, rather than just the mere collection of members.
- (2) **Power**, according to the family system refers to the messages and rules that shape members. They induce guilt, control or limit behaviour; they preserve themselves from extinction, being repetitive and abundant. They are considered to be the relationship agreements that prescribe and limit family members' behaviour over time (Morgaine, 2001). In a well-functioning family, a clear hierarchical power exists, in which individuals share power in decision-making, and the younger generation has what is referred to as "equal overt power" (Walsh, 2003). This implies that individual choice is given, working towards a goal of family members speaking up and being respected as significant individuals who can make valuable contributions in decision-making (Walsh, 2003).
- (3) **Structures** in a family can be created by modifications that can occur through learning, events, change or through positive feedback loops (Oetter & StevensDomiquez, 1998). The authors also differentiate between first order and

second order changes. First order changes are described as minor structural changes that create vulnerability within the system, while the system stays the same. Second order changes are high level changes that could cause a change within the systems structure. Feedback loops (Oetter & StevensDomiquez, 1998) are viewed as pathways for information sharing, starting from one point of the system, through other parts and back to the point of origin. There are 2 forms of feedback loops: positive feedback, which play a vital role in creating and maintaining the stability of a system and negative loops that function to restore and maintain the equilibrium of the system.

- (4) Boundaries are the emotional barriers that protect and further improve the morals and values of the system (Oetter & StevensDominquez, 1998). According to Morgaine (2001), boundaries can be viewed in a series of similarities, from open to closed system, to establish a clear line between those inside and those outside of the system. Open boundaries allow elements outside of the family to have an influence. Closed boundaries involve isolation of members from the environment, being self-contained. No family, however, as Morgaine (2001) informs, is considered to be entirely open or closed systems. According to Walsh (2003), well-functioning families possess enough strength and integrity to allow active involvement within its borders and through this interaction, bring varied interests and excitement back into the family.
- (5) **Family Roles**, defined by Miller, Ryan, Keitner, Bishop & Epstein (2000) & Walsh (2003) as the recurrent patterns of behaviour by which family functions are to be fulfilled. They further divide family roles into two areas: *instrumental* and *affective*, and further subdivide these into two more spheres: *necessary family functions* (functions with which the family must be concerned) and *other family functions*

(functions that are not necessary for effective functioning, yet arise in the life of every family). Walsh, (2003) describes two additional and important aspects of role functioning: *role allocation* (referring to the families patterns of assigning roles to members) and *role accountability* (referring to the procedures within the family to ensure that these assigned roles are fulfilled by its members.

(6) **Communication patterns.** Family communication is defined as the act of making information, ideas, thoughts and feelings known, referring to the verbal (i.e. spoken words) and non-verbal (i.e. facial expression, eye contact, movement, etc.) behaviour of family members. It assists a family to alter their cohesion and flexibility (adaptability by Buswell et al., 2012) to meet their developmental and situational demands (Smith et al., 2009). Family communication is a facilitating dimension for cohesion and adaptability. Cohesion, according to Buswell et al. (2012), is defined as a feeling of emotional closeness with another person. This is ultimately referred to as 'a sense of belonging'. According to Olsen (2000), emotional bonding, boundaries, coalitions, time, space, friends, decision-making, interest and recreation are those variables that can be used to diagnose family cohesion. Olsen (2000) and Sturge-Apple et al. (2010) further identify four levels of cohesion: disengaged (very low); separated (low to moderate); connected (moderate to high); and enmeshed (very high) levels of emotional closeness. Separated and connected levels of cohesion imply that the individuals experience a balance and are able to be both independent from, and connected to, the family. Enmeshed systems imply much consensus and less independence. Disengaged systems refer to limited attachment or commitment to family. Adaptability refers to the ability of the family to change power structures, roles and rules within the relationship. This framework implies that family cohesion and family adaptability are characteristics or components of healthy family functioning (Buswell et al., 2012).

Family system theory focuses on family dynamics, such as; relationships; power; structures; boundaries; family roles; and communication patterns; therefore, family behaviour can be understood by viewing the family as a unit, rather than just its individual facets (Buswell et al., 2012).

6.7. Limitations of the study

No research is without its limitations. For this study the limitations were as follows:

- The study was conducted using self-reported questionnaires that might have indicated a measure of truth but not the actual truth, as adolescent participants were required to only indicate *their own perceptions* of their family functioning, decision-making abilities and decision-making styles, as *they* understood it. Other role players were not involved to compare these findings in a holistic manner.
- The study sample used, did allow generalization of these findings across all racial groups and research population, since random sampling was followed which represents the racial distribution within schools.

6.8. Recommendations

• There is a need for workshops and training on the process of decision-making and decision-making styles for adolescents, as well as, general advice and information about the adolescence phase that every individual experiences. Fostering adolescents' self-endorsed functioning through empathy, giving choice wherever possible, and

encouraging them to act upon their personal values and interests, seems crucial in order to successfully deal with the challenges of adolescence (Van Pentegem et al., 2012).

- The concept of weakened families created by the democratic and other related conditions in this country has caused disparity between idealization and the reality of the family, as often, living conditions disadvantage beliefs and perceptions regarding what families are and what they should be. This identifies a need for family preservation services in the form of training for parents, as well as, family communication skills training.
- These findings may be needed by Departments such as Social Development and other family and child care organizations for working with families with adolescents in South Africa. With the greatest aim of working in a preventative manner/ early intervention. Social workers and other child care practitioners may benefit in this that they may have some guidelines onto service delivery in this field.
- The results of this study may be used in the child and adolescent field of studies still to be
 done and various other welfare and health professionals in dealing with adolescents and
 their decision making abilities within a family context.

6.9. Conclusion

The main results on the relationship between variables of this research study indicate no positive relationship between the dimensions of family functioning and the decision-making styles of adolescents. There is however significant negative relationships found between Communication and Hyper vigilance decision making style. The Hyper vigilance style refers to the application of an optimistic views towards the various alternatives of a decision needed to be made, panic and stress sets in and an objectionable alternative is considered. Another significant negative relationship was found between Roles and Buck-passing decision making

style. The buck-passing decision-making style refers to the passing of a decision on to someone else

With adolescence considered to be a crucial and very unique developmental period and a fundamental phase of growing up. They desire independence and independent decisionmaking which, however, requires on-going learning and instruction. Family is considered to be crucial to all its members and, based on the challenges that adolescents have to face; the recommendation is that adolescence be viewed as *ultimately* important to the family as well. Adolescence can invite change by influence. Certain South African children may be given an opportunity to prosper, if they were to be given the chance to experience stability within their families and/or communities. Further development in decision-making abilities would enhance their problem-solving abilities and self-esteem. With the relevant skills training on the several dimensions of families and family functioning, adolescents become more familiar with the importance of effective decision-making and how it affects their lives and the environment. Adolescents also become more aware of how their decisions (good or bad) impact on the other family members, in whichever roles they occupy within the family system. Parents may become aware of the importance of adolescents' independent decisionmaking as well, and endeavor to guide them to make good choices in life. The writer trusts that the results of this study would act as a guide and framework for effective programmes that could be implemented for adolescents.

REFERENCES

Aarons, G., McDonald, E.J., Connelly, C. & Newton, R. (2007). Assessment of Family Functioning in Caucasian and Hispanic Americans: Reliability, Validity, and Factor Structure of the Family Assessment Device. Family Process. 46:557–569.

Akister, J. & Stevenson-Hinde, J. (1991). Identifying families at risk: Exploring the Potential of the McMaster Family Assessment Device. Journal of Family Therapy, 13: 411-421.

Albert, S. & Steinberg, L. (2011). Judgment and decision making in adolescence. Journal of research on adolescence, 21(1), 211-224.

Alloy, L.B., Zhu, L. & Abramson, L.Y. Cognitive vulnerability to depression: implications for adolescent risk behavior in general. in: D. Romer (Ed.) Reducing adolescent risk: toward an integrated approach. Sage Publications, Thousand Oaks (CA); 2003:171–182.

Amoateng, A.Y., Richter, L. M., Makiwane, M. & Rama, S. (2004). Describing the structure and needs of families in South Africa: Towards the development of a national policy framework for families. A report commissioned by the Department of Social Development. Pretoria: Child Youth and Family Development, Human Sciences Research Council.

Anastasi. A. (1982). Psychological Testing: Basic Concepts and Common Misconceptions

Armesh, H. (2013) Decision Making. Multi Media University Malysia.

Arthur, J., Waring, M., Coe, R. & Hedges, L.V. (2012) Research methods & methodologies in Education: Sage publications, 370.721 RES.

Aufseeser, D., Jekielek, S. & Brown, B. (2006). The family environment and adolescent well-being: Exposure to positive and negative family influences: Child trends and the National Adolescent Health Information Centre.

Babbie, E. & Mouton, J. (2010). The practice of social research. Cape Town: Oxford University Press.

Bandura, A. (2005). Chapter 1. Adolescent Development from agentic perspective. Self-Efficacy beliefs of adolescent.1-43

Berzonsky, M.D. & Ferrari, J.R. (1996). Identity orientation and decisional strategies. (20)5, 597-606.

Bogenschneider, K. (2006). Teaching family policy in undergraduate and graduate classrooms: Why it's important and how to do it better. Family Relations, 55, 16-28.

Brown, J., Abdallah, S.S. & Ng, R. (2011). Decision making styles East and West: Is it time to move beyond cross-cultural research? International Journal of Sociology and Anthropology, 3(12), 452-459,

Burnett, P.C. (1991). Decision making style and self-concept. Australian Psychologist, 26(1), 55-58.

Buswell, L., Zarbriskie, R.B., Lundberg, N. & Hawkins, A.J. (2012). The Relationship between father involvement in family leisure and family functioning: The importance of daily leisure time. Leisure Sciences, 34, 172-190.

Carpenter, B. & Mulligan, E. (2009). Family, Know Thyself: A Workbook Based Intergenerational Intervention to Improve Parent Care Coordination: Clinical Gerontologist, 32: 147-163.

Cenkseven-Onder, F. (2012). The influence of decision making styles on early adolescent life satisfaction. Social Behavior and Personality, 40(9), 1523-1536.

Cox, M.J. & Paley, B. (1997). Families as systems. Annual Rev. Psychology. 48:243-67.

Creswell, J.W. (2009). Research Design: Qualitative, Quantitative & mixed methods. 3rd edition. Los Angeles. Sage Publications.

David, C. (2010). Quantitative Psychological Research. The complete students' companion, 3rd Edition: Psychology Department, Staffordhire University: Psychology Press.

Deniz, E.M. (2006). The relationship among coping with stress, life satisfaction, decision-making styles and decision self-esteem: An investigation with Turkish University Students. Social Behaviour and Personality, 34(9), 1161-1170.

Deniz, M.E. (2011). An Investigation of Decision Making Styles and the Five-Factor Personality Traits With Respect To Attachment Styles Educational Sciences: Theory & Practice, 11(1), 105-113.

Dewberry, C., Juanchich, M. & Narendran, S. (2013). Decision-making competence in everyday life: The roles of general cognitive styles, decision-making styles and personality. Personality and Individual Differences. 55. 783–788.

Eccles, E. & Gootman J. (2002). Community Programs to Promote Youth Development. National Academy.

Epstein, N.B., Baldwin, L.M. & Bishop, D.S. (1983). The McMaster Family Assessment Device. Journal of Marital and Family Therapy, 9(2), 171-180.

Fantasia, H.C. (2008). Concept analysis: Sexual Decision-Making in Adolescence. Nursing Forum, 43(2).

Frank, S.J., Pirsch, L.A. & Wright, V.C. (1990). Late adolescents' perceptions of their relationships with their parents: Relationships among deidealization, autonomy, relatedness, and insecurity and implications for adolescent adjustment and ego identity status. Journal of Youth and Adolescence, 19(6).

Galotti, K.M., Ciner, E., Altenbaumer, H.E., Geerts, H.J., Rupp, A. & Woulfe, J. (2006). Decision making styles in a real-life decision: choosing a college major. Personality and Individual Differences. 41:629-639.

Graham, A. & Graham, C. (2001). Families, households and society. Palgrave: Gray, D.E. (2009). Doing Research in the Real World, 2nd Edition. *Sage Publications*.

Henderson, C.J., Dakof, G.A., Schwartz, S.J. & Liddle, H.J. (2006). Family Functioning, Self-Concept, and Severity of Adolescent Externalizing Problems: Sage Publications.

Hochfeld, T. (2007). Missed opportunities: conservative discourses in the draft National Family Policy of South Africa. International Social Work, 50(1):79-91.

Holborn, L. & Eddy, G. (2011). First steps to healing the South African family. South African Institute of Race Relations. South African Family.

Howitt, D. & Cramer, D. (2000). First Steps in Research and Statistics, A practical Workbook for Psychology students: London and Philadelphia: Routledge Publishers.

Human-Hendrick, A.R. & Roman, N.V. (2014). What is the link between antisocial behaviour of adolescents and parenting. A systematic review of parenting practices to manage antisocial behaviour: Antisocial Behaviour.

Isen, A.M. (2001). An Influence of Positive Affect on Decision Making in Complex Situations: Theoretical Issues with Practical Implications. Journal of Consumer Psychology, (11)2, 75-85.

Jackson, S.A. & Kleitman, S. (2014). Individual differences in decision making and confidence. Capturing decision tendencies in a fictitious medical test. Metacognition and Learning, 9(1), 25-49.

Jobe-Shields, L., Buckholdt, K.E., Parra, G.R. & Tillery, R.N. (2014). Adolescent reactions to maternal responsiveness and internalizing symptomatology: A daily diary investigation. Personal Relationships, 21, 335–348.

Johnson, N., Frenn, M., Feetham, S. & Simpson, P. (2011). Autism Spectrum Disorder: Parenting stress, Family Functioning and Health Related Quality of Life. Families, Systems and Health, 29(3), 232-252.

Jurisova, E. & Sermany-Schuller, I. (2013). Structure of basal physical regulations and personality integration in relation to coping strategies in decision making in paramedics.

Studia Psychologica, 55(1).

Kambam, P. & Thompson, C. (2009). The development of decision making capacities in children and adolescents: Psychosocial and Neurological perspectives and their implications for juvenile defendants. Behavioural sciences and the law, 27, 173-190.

Kamaruddin, A. R. & Mokhlis, S. (2003). Consumer socialization, social structural factors and decision-making styles: a case study of adolescents in Malaysia. International Journal of Consumer Studies, 27: 2, 145–156.

Koen, V., Van Eeden, C. & Venter, C. (2011). African female adolescents' experience of parenting and their sense of well-being. Journal of Psychology in Africa, 21(2), 197-210.

Levinson, B.A.U., Sutton, M. & Winstead, T. (2009). Education Policy as a practice of power. Theoretical tools, ethnographic methods, demographic options. Educational Policy. (23)6. 767-795.

Lizarraga, M., Baquendano, M., Oliver, M. & Closas, A. (2009). Development and validation of a decision making questionnaire. Journal of Guidance and Counselling, 37(3), 257-373.

Louw, D.A., Van Ede, D. M., & Louw, A. E. (1998). Human development (2nd ed.). Pretoria: Kagiso.

Luyckx, K., Soenens, B., Berzonsky, M.D., Smits, I., Goossens, L. & Vansteenkiste, M. (2007). Information-oriented identity processing, identity consolidation, and well-being: The moderating role of autonomy, self-reflection, and self-rumination. Personality and Individual Differences, 43, 1099–1111.

Makiwane, M. & Kwizera, S. (2008) Youth and Well-being: a South African case study. Social indicators research.

Makiwane, M., Makoae, M., Botsis, H. & Vawda, M. (2012). A baseline study on families in Mpumalanga. Human Sciences Research Council.

Mann, L., Burnett, P., Radford, M. & Ford, S. (1997). The Melbourne Decision Making Questionnaire: An instrument for measuring patterns for coping with decisional conflict. Journal of Behavioral Decision Making, 10(1), 1-19.

Marriages Act No 25 of 1961. Pretoria: Government Printers. South Africa. (1961).

Martinez, L. (2007). Sensory evaluation based on linguistic decision analysis. Journal of Approximate Reasoning, 44, 148–164.

McCreary, L.L. & Dancy, B.L. (2004). Dimensions of Family Functioning: Perspectives of Low-Income African American Single-Parent Families. Journal of Marriage and Family. 66: 690–701.

McGrath, R.E. (2011). Quantitative Methods in Psychology.

WESTERN CAPE

McKinney C., Donnelly R. & Renk, K. (2008). Perceived Parenting, Positive and Negative Perceptions of Parents and Late Adolescent Emotional Adjustment: Child and Adolescent Mental Health: Child and Adolescent Mental health, 13(2), 66-73.

Miley, K.K., O'Melia, M. & DuBois, B. (2009). Generalist social work practice: An empowering approach. 6th ed. Boston: Allyn & Bacon.

Miller, D.C. & Byrnes, J.P. (2001). To achieve or not to achieve: A self-regulation Perspective on adolescent's academic decision making: Journal of educational Psychology, 93(4), 677-685.

Miller, I.W., Ryan, C.E., Keitner, G.I., Bishop, D.S. & Epstein, N.B. (2000). "Factor Analyses of the Family Assessment Device," by Ridenour, Daley, & Reich. Family Process. 39:141-144.

Miller, I.W., Ryan, C.E., Keitner, G.L., Bishop, D.S. & Epstein, N.B. (2000). The McMaster Approach to Families Theory, assessment, treatment and research: The Association for Family Therapy and Systematic Practice: Journal of Family Therapy, 22, 168-189.

Morgaine, C. (2001). Family Systems Theory. CFS410U.

Muncie, J. & Sapsford, R. (1995). Chapter 1. Issues in the study of "The Family". Understanding the Family. Sage Publications.

Neale, B. (2000). Theorizing family, kinship and social change. Workshop paper 6. Prepared for workshop two: statistics and theories for understanding social change.

Oetter, P. & StevensDominquez, M. (1998) Performance Competence. Lifespan and Framework. Environmental Influences and the Individual.

Ogwo, A. (2013). Adolescents-Parent Relationships as Perceived By Younger and Older Adolescents. Department of Psychology. Nasarawa State University.

Olsen, D.H. (2000). Circumplex Model of Family and Family Systems. Journal of Family Therapy, 22: 144-167.

Parker, M. & Fischhoff, B. (2005). Decision-making Competence: External Validation through an Individual-differences Journal of Behavioral Decision Making. 18: 1–27.

Parra, A., Oliva, A. & Reina, M.C. (2013). Family relationships from adolescence to emerging adulthood: A longitudinal Study. Journal of Family Issues, 1–19.

Petrocelli, J.V., Calhoun, G.B. & Glaser, B.A. (2003). The Role of General Family Functioning in the Quality of the Mother-Daughter Relationship of Female African American Juvenile Offenders. Journal of Black Psychology, (29) 4.378-392.

Randolph, S.M. & Hassan, F.M.A. (1996). Structural Adjustment and Family Policy in Africa: Lessons from Sudan. Journal of Family and Economic Issues, 17(2).

Robila, M. (2011). International perspectives on family policies. Child and family studies, 21: 1-3.

Roche, K.M., Ensminger, M.E. & Cherlin, A.J. (2007). Variations in parenting and adolescent outcomes among African American and Latino families living in low-income, urban areas. Journal of family issues.(28)7:882.

Roman, N.V. (2014). Parenting in a rainbow Nation: A South African perspective on parenting. Child and Family Studies.

Roman, N. & Davids, E. (2013). "To do or not to do". Examining decision making styles of university students in South Africa: Comparing family structure, gender and living arrangements. Decision making of young adults.

Roman, V.V. & Frantz, J.M. (2013). The prevalence of intimate partner violence in the family: A systematic review of the implications for adolescents in Africa. Family Practice.

Roman, N.V., Human, A. & Hiss, D. (2012). Young South African adults' Perceptions of Parental psychological control and anti-social behaviour. Social Behaviour and Personality, (40)7, 1163-1174.

Romich, J.L., Lundberg, S. & Tsang, K.P. (2009). Independence giving or autonomy granting? Childhood predictors of decision sharing patterns between young adolescents and parents. Journal of research on adolescence, 19(4), 587-600.

WESTERN CAPE

Ryan, C.E., Epstein, N.B., Keitner, G.I., Miller, I.W. & Bishop, D.S. (2005). Evaluating and treating families: the McMaster Approach. Routledge

Saunders, P. (1999). Australian Institute of Family Studies. Research plan 1999 – 2001. Commonwealth of Australia. Routledge.

Seiffge-Krenke, I. (2002). "Come on, say something, Dad!". Communication and coping fathers of diabetic adolescents. Journal of Paediatric Psychology, 5(27), 439-450.

Sevenhuijsen, S., Bozalek, V., Gouws, A. & Minnaar-Mcdonald, M. (2003). South African social welfare policy: an analysis using the ethic of care. Critical Social Policy, 23(3), 299-321.

Shek, D. (2002). Family Functioning and Psychological Wellbeing, Social Adjustment and Problem Behavior in Chinese Adolescents with and without Economic Disadvantage: The Journal of Genetic Psychology, 163(4), 497-502.

Sheriif, B., Seedat, M. & Saffla, S. (2010). A critical review of family functioning indices. African Safety Promotion Journal. 8:2.

Siegel, D.J. & Hartzel, M. (2014). Parenting from the inside out. How a deeper self-understanding can help you raise children who thrive. 10th Anniversary edition. Penguin Publishing. Family and Relationships.

Smetana, J., Crean, H.F. & Campione-Barr, N. (2005). Adolescents' and parents' changing conceptions of parental authority. New Directions for Child and Adolescent Development, 108.

Smith, K.M., Freeman, P.A. & Zabriskie, R.B. (2009). An examination of family communication within the core balance model of family leisure functioning. Family relations, 58, 79-90.

Smith, D.G., Xiao, L. & Bechara, A. (2012) Decision making in children and and adolescents: Impaired Iowa Gambling task performance in early adolescence. Development Psychology. (48)4. 1180-1187.

Soenens, B. & Vansteenkiste, M. (2009). A theoretical upgrade of the concept of parental psychological control: proposing new insights on the basis of self-determination theory. Developmental Review. (30). 74-99.

Stattin, H., Persson, S., Burk, W.J. & Kerr, M. (2011). Adolescents' perceptions of the democratic functioning in their families. Europeun Psychologist, 16(1), 32-42.

Steinberg L. (2005). Cognitive and affective development in adolescence, 9(2), 69-74.

Sturge-Apple, M.L., Davies, P.T. & Cummings, E.M. (2010). Typologies of family functioning and childrens adjustment during the early school years. Child Development. (81)4:1320-1335.

Sussman, M.B., Steinmetz, S.K. & Peterson, G.W. (1999). Handbook of marriage and the family. 2nd ed. New York, Plenum Press.

Tafa, M. & Baiocco, R. (2009). Addictive behaviour and family functioning during adolescence. The American journal of family therapy, 37; 388-395.

The White Paper on families in South Africa. (2013). Department of Social Development South Africa.

Van Pentegem, S., Beyers, W., Vansteenkiste, M. & Soenens, B. (2012). On the association between adolescent autonomy and psychosocial functioning: examining decisional independence from a self-determination theory perspective. Developmental Psychology. 48(1), 76-88.

Walsh, F. (2003). Normal family processes, Third Edition, Growing diversity and complexity.

Wray-Lake, L., Crouter, A.C. & McHale, S.M. (2010). Developmental patterns in decision-making autonomy across middle childhood and adolescence: European American parents' perspectives. Child Development, 81(2), 636-651.

Yamane, T. (1967). Statistics: An Introductory Analysis, 2nd Edition, New York: Harper and Row.

Zarrett N. & Eccles J. (2006). The passage to adulthood challenges of late adolescence, New Directions for Youth Development, 111.

Zuberi, A. (2013). Neighbourhoods and parenting: Assessing the influence of neighbourhood quality on the parenting monitoring of youth. Youth & Society. 1-30.

APPENDICES

Appendix I: Questionnaire

Questionnaire

Please complete the following by circling the	correct response.
---	-------------------

Section A

Gender	Male					Female			
Age	Terriale Terriale								
Grade									
Grade			ı				1		
Race	Coloured Black / A		Black / A	frican V		/hite	Indian / Asian		
Home language	Afrikaans	Afrikaans Englis		sh	isiXhosa		Other		
Who do you live with	Both		Mother	Father Only		Caregiver /		Alone	
	Parents		Only			Guardian			
Are your parents	Married		Living	Singl	e, do	Single		Single	
		t	ogether	not live		because he /		because he	
	9	\leq	but not	together		r she is		/ she is	
			married	and are not		ot widowed		divorced	
				married					

Section B: Decision Making Styles

The next section consists of two parts, Part I and II, that looks at the decisions you make.

WESTERN CAPE

Part I:

Instructions

People differ in how comfortable they feel about making decisions. Please indicate how you feel about making decisions by ticking the response which is most applicable to you.

		True for me	Sometimes true	Not true for me
(1)	I feel confident about my ability to make decisions	[]	[]	[]
(2) (3)	I feel inferior to most people in making decisions I think that I am a good decision maker	[] []	[]	[]
(4)	I feel so discouraged that I give up trying to make decisions	[]	[]	[]
(5)	The decisions I make turn out well	[]	[]	[]

(6)	It is easy for other people to convince me that							
	their decision rather than mine is the correct	[]	[]	[]				
	one							

PART II

Instructions:

People differ in the way they go about making decisions. Please indicate how you make decisions by ticking for each question the response which best fits your usual style.

When making decisions -		True for me		Sometime s true			Not true for me	
1.	I feel as if I'm under tremendous time pressure when making decisions	[]	[]	[]	
2.	I like to consider all of the alternatives	[]]]	[]	
3.	I prefer to leave decisions to others]	[]	[]	
4.	I try to find out the disadvantages of all alternatives	1]]]	[]	
5.	I waste a lot of time on trivial matters before getting to the final decision	of the]	[]	[]	
6.	I consider how best to carry out the decision	[]	[]	[]	
7.	Even after I have made a decision I delay acting upon it	[]	[]	[]	
8.	When making decisions I like to collect lots of information	[]]]	[]	
9.	I avoid making decisions	[]	[]	[]	
10.	When I have to make a decision I wait a long time before starting to think about it	[]]]	[]	
11.	I do not like to take responsibility for making decisions	[]	[]	[]	
12.	I try to be clear about my objectives before choosing	[]	[]	[]	

13.	The possibility that small things might go wrong causes me to swing abruptly in my preferences	[]	[]	[]
14.	If a decision can be made by me or another person I let the other person make it	[]]]	[]
Whe	en making decisions -		e for ne		etime rue	Not for	
15.	Whenever I face a difficult decision I feel pessimistic about finding a good solution	[]	[]	[]
16.	I take a lot of care before choosing	[]	[]	[]
17.	I do not make decisions unless I really have to	[]	[]	[]
18.	I delay making decisions until it is too late]	[]	[]
19.	I prefer that people who are better informed decide for me	1]	[]	[]
20.	After a decision is made I spend a lot of time convincing myself it was correct	of the]	[]	[]
21.	I put off making decisions	[]	[]	[]
22.	I cannot think straight if I have to make decisions in a hurry	[]	[]	[]

Section C: Parenting Styles

The following questions are about the attitudes and behaviours of your parents or guardians. If you stay with someone other than your mother, who is a female still complete the mother / female parenting form, if you stay with someone other than your father, who is male still complete the father / male form. If you do not stay with both your parents / guardians complete only the relevant form and leave the other section blank.

MOTHER / FEMALE PARENTING FIGURE FORM

This questionnaire lists various attitudes and behaviours of parents. As you remember your MOTHER would you place a tick in the most appropriate box next to each question.

Was responsive to my feelings or needs	Not at all like her	Not like her	Somewhat like her	A lot like her
Used physical punishment as a way of disciplining me.	Not at all like her	Not like her	Somewhat like her	A lot like her
Took my desires into account before asking me to do something.	Not at all like her	Not like her	Somewhat like her	A lot like her
When I asked why I had to conform, [she stated]: because I said so, or I am your parent and I want you to.	Not at all like her	Not like her	Somewhat like her	A lot like her
Explained to me how she felt about my good and bad behaviour.	Not at all like her	Not like her	Somewhat like her	A lot like her
Spanked me when I was disobedient.	Not at all like her	Not like her	Somewhat like her	A lot like her
Encouraged me to talk about my troubles.	Not at all like her	Not like AP her	Somewhat like her	A lot like her
Found it difficult to discipline me.	Not at all like her	Not like her	Somewhat like her	A lot like her
Encouraged me to freely express myself even when I disagreed with them.	Not at all like her	Not like her	Somewhat like her	A lot like her
Punished me by taking privileges away from me with little if any explanations.	Not at all like her	Not like her	Somewhat like her	A lot like her
Emphasized the reasons for rules.	Not at all like her	Not like her	Somewhat like her	A lot like her
Gave comfort and understanding when I was upset.	Not at all like her	Not like her	Somewhat like her	A lot like her
Yelled or shouted when I misbehaved.	Not at all like her	Not like her	Somewhat like her	A lot like her
Gave praise when I was good.	Not at all like her	Not like her	Somewhat like her	A lot like her

Gave into me when I caused a commotion about something.	Not at all like her	Not like her	Somewhat like her	A lot like her
Exploded in anger towards me.	Not at all like her	Not like her	Somewhat like her	A lot like her
Threatened me with punishment more often than actually giving it.	Not at all like her	Not like her	Somewhat like her	A lot like her
Took into account my preferences in making plans for the family.	Not at all like her	Not like her	Somewhat like her	A lot like her
Grabbed me when I was being disobedient.	Not at all like her	Not like her	Somewhat like her	A lot like her
Stated punishments to me and did not actually do them.	Not at all like her	Not like her	Somewhat like her	A lot like her
Showed respect for my opinions by encouraging me to express them.	Not at all like her	Not like her	Somewhat like her	A lot like her
Allowed me to give input into family rules.	Not at all like her	Not like her	Somewhat like her	A lot like her
Scolded and criticized me to make me improve.	Not at all like her	Not like her	Somewhat like her	A lot like her
Spoiled me.	Not at all like her	Not like AP her	Somewhat like her	A lot like her
Gave me reasons why rules should be obeyed.	Not at all like her	Not like her	Somewhat like her	A lot like her
Uses threats as punishment with little or no justification.	Not at all like her	Not like her	Somewhat like her	A lot like her
Had warm and intimate times together with me.	Not at all like her	Not like her	Somewhat like her	A lot like her
Punished me by putting me off somewhere alone with little if any explanations.	Not at all like her	Not like her	Somewhat like her	A lot like her
Helped me to understand the impact of my behaviour by encouraging me to talk about the consequences of my own actions.	Not at all like her	Not like her	Somewhat like her	A lot like her
Scolded and criticized me when my behaviour didn't meet their expectations.	Not at all like her	Not like her	Somewhat like her	A lot like her

Explained the consequences of my behaviour.	Not at all like her	Not like her	Somewhat like her	A lot like her
Slapped me when I misbehaved.	Not at all like her	Not like her	Somewhat like her	A lot like her

FATHER / MALE PARENTING FIGURE FORM

This questionnaire lists various attitudes and behaviours of parents. As you remember your FATHER would you place a tick in the most appropriate box next to each question

Was responsive to my feelings or needs	Not at all like him	Not like him	Somewhat like him	A lot like him
Used physical punishment as a way of disciplining me.	Not at all like him	Not like him	Somewhat like him	A lot like him
Took my desires into account before asking me to do something.	Not at all like him	Not like him	Somewhat like him	A lot like him
When I asked why I had to conform, [he stated]: because I said so, or I am your parent and I want you to.	Not at all like him	Not like him	Somewhat like him	A lot like him
Explained to me how he felt about my good and bad behaviour.	Not at all like him	Not like him	Somewhat like him	A lot like him
Spanked me when I was disobedient.	Not at all like him	Not like him	Somewhat like him	A lot like him
Encouraged me to talk about my troubles.	Not at all like him	Not like him	Somewhat like him	A lot like him
Found it difficult to discipline me.	Not at all like him	Not like him	Somewhat like him	A lot like him
Encouraged me to freely express myself even when I disagreed with them.	Not at all like him	Not like him	Somewhat like him	A lot like him
Punished me by taking privileges away from me with little if any explanations.	Not at all like him	Not like him	Somewhat like him	A lot like him
Emphasized the reasons for rules	Not at all like him	Not like him	Somewhat like him	A lot like him
Gave comfort and understanding when I was upset.	Not at all like him	Not like him	Somewhat like him	A lot like him

Yelled or shouted when I misbehaved.	Not at all like him	Not like him	Somewhat like him	A lot like him
Gave praise when I was good.	Not at all	Not like	Somewhat	A lot like
	like him	him	like him	him
Gave into me when I caused a commotion about something.	Not at all like him	Not like him	Somewhat like him	A lot like him
Exploded in anger towards me.	Not at all	Not like	Somewhat	A lot like
	like him	him	like him	him
Threatened me with punishment more often than actually giving it.	Not at all	Not like	Somewhat	A lot like
	like him	him	like him	him
Took into account my preferences in making plans for the family.	Not at all	Not like	Somewhat	A lot like
	like him	him	like him	him
Grabbed me when I was being disobedient.	Not at all	Not like	Somewhat	A lot like
	like him	him	like him	him
Stated punishments to me and did not actually do them.	Not at all	Not like	Somewhat	A lot like
	like him	him	like him	him
Showed respect for my opinions by encouraging me to express them.	Not at all	Not like	Somewhat	A lot like
	like him	him	like him	him
Allowed me to give input into family rules.	Not at all like him	Not like	Somewhat like him	A lot like him
Scolded and criticized me to make me improve	Not at all like him	Not like him	Somewhat like him	A lot like him
Spoiled me.	Not at all	Not like	Somewhat	A lot like
	like him	him	like him	him
Gave me reasons why rules should be obeyed.	Not at all	Not like	Somewhat	A lot like
	like him	him	like him	him
Uses threats as punishment with little or no justification.	Not at all like him	Not like him	Somewhat like him	A lot like him
Had warm and intimate times together with me.	Not at all	Not like	Somewhat	A lot like
	like him	him	like him	him
Punished me by putting me off somewhere alone with little if any explanations.	Not at all	Not like	Somewhat	A lot like
	like him	him	like him	him

Helped me to understand the impact of my behaviour by encouraging me to talk about the consequences of my own actions.	Not at all	Not like	Somewhat	A lot like
	like him	him	like him	him
Scolded and criticized me when my behaviour didn't meet their expectations.	Not at all	Not like	Somewhat	A lot like
	like him	him	like him	him
Explained the consequences of my behaviour.	Not at all	Not like	Somewhat	A lot like
	like him	him	like him	him
Slapped me when I misbehaved.	Not at all like him	Not like him	Somewhat like him	A lot like him

Section D: Family Functioning

Here are a number of statements about families. Read each statement carefully, and decide how well it describes your own family. There are no right or wrong answers. Try not to spend too much time thinking about each statement, but respond as quickly and as honestly as you can. If you have difficulty, answer with your first reaction.

Use the following scale to respond:

- 1 (Strongly agree): if you feel the statement describes your family very accurately
- 2 (Agree): if you feel the statement describes your family for the most part
- 3 (Disagree): if you feel the statement does not describe your family for the most part
- 4 (Strongly disagree): if you feel the statement does not describe your family at all

WESTERN CAPE					
1 (Strongly Agree)	2 (Agree)	3 (Disagree)	4 (Strongly disagree)		

Planning family activities is difficult because we misunderstand each other. We resolve most everyday problems around the house. When someone is upset the others know why. When you ask someone to do something, you have to check that they did it. If someone is in trouble, the others become too involved. In times of crisis we can turn to each other for support. We don't know what to do when an emergency comes up. We sometimes run out of things that we need. We are reluctant to show our affection for each other. We make sure members meet their family responsibilities. We cannot talk to each other about the sadness we feel.		
When someone is upset the others know why. When you ask someone to do something, you have to check that they did it. If someone is in trouble, the others become too involved. In times of crisis we can turn to each other for support. We don't know what to do when an emergency comes up. We sometimes run out of things that we need. We are reluctant to show our affection for each other. We make sure members meet their family responsibilities.	Planning family activities is difficult because we misunderstand each other.	
When you ask someone to do something, you have to check that they did it. If someone is in trouble, the others become too involved. In times of crisis we can turn to each other for support. We don't know what to do when an emergency comes up. We sometimes run out of things that we need. We are reluctant to show our affection for each other. We make sure members meet their family responsibilities.	We resolve most everyday problems around the house.	
If someone is in trouble, the others become too involved. In times of crisis we can turn to each other for support. We don't know what to do when an emergency comes up. We sometimes run out of things that we need. We are reluctant to show our affection for each other. We make sure members meet their family responsibilities.	When someone is upset the others know why.	
In times of crisis we can turn to each other for support. We don't know what to do when an emergency comes up. We sometimes run out of things that we need. We are reluctant to show our affection for each other. We make sure members meet their family responsibilities.	When you ask someone to do something, you have to check that they did it.	
We don't know what to do when an emergency comes up. We sometimes run out of things that we need. We are reluctant to show our affection for each other. We make sure members meet their family responsibilities.	If someone is in trouble, the others become too involved.	
We sometimes run out of things that we need. We are reluctant to show our affection for each other. We make sure members meet their family responsibilities.	In times of crisis we can turn to each other for support.	
We are reluctant to show our affection for each other. We make sure members meet their family responsibilities.	We don't know what to do when an emergency comes up.	
We make sure members meet their family responsibilities.	We sometimes run out of things that we need.	
	We are reluctant to show our affection for each other.	
We cannot talk to each other about the sadness we feel.	We make sure members meet their family responsibilities.	
	We cannot talk to each other about the sadness we feel.	
We usually act on our decisions regarding problems.	We usually act on our decisions regarding problems.	

You only get the intere	st of others when sc	mething is impor	tant to them.		
You can't tell how a pe	rson is feeling from	what they are say	ing.		
Family tasks don't get s	spread around enou	gh.			
Individuals are accepte	d for what they are.				
You can easily get away	y with breaking the r	rules.			
People come right out	and say things instea	ad of hinting at th	em.		
Some of us just don't re	espond emotionally.				
We know what to do in	an emergency.				
We avoid discussing ou	ir fears and concern:	S.			
1 (Strongly Agree)	2 (Agree)	3 (Disagree)	4 (Strongly disagree)		
It is difficult to talk to e	ach other about ten	der feelings.			
We have trouble meeti	ng our financial obli	gations.			
After our family tries to not.	solve a problem, w	e usually discuss	whether it worked or		
We are too self-centred	d	Ш_Ш_Ш_			
We can express feeling	s to each other.	ERSITY of the			
We have no clear expe	ctations about toilet	habits.			
We do not show our lo	ve for each other.				
We talk to people directly rather than through go-betweens.					
Each of us has particular duties and responsibilities.					
There are lots of bad feelings in the family.					
We have rules about hitting people.					
We get involved with each other only when something interests us.					
There is little time to explore personal interests.					
We often don't say what we mean.					
We feel accepted for w	hat we are.			<u> </u>	
We show interest in ea	ch other when we c	an get something	out of it personally.	<u> </u>	
We resolve most emotional upsets that come up.					
Tenderness takes second place to other things in our family.					
We discuss who are responsible for household jobs.					
Making decisions is a problem for our family.					

Our family shows interest in each other only when they can get something out of it.				
We are frank (direct, str	aightforward) with ea	ach other.		
We don't hold to any rul	es or standards.			
If people are asked to do	something, they nee	ed reminding.		
We are able to make de	cisions about how to	solve problems.		
If the rules are broken, v	ve don't know what t	o expect.		
Anything goes in our fan	nily.			
We express tenderness.				
We confront problems in	nvolving feelings.			
1 (Strongly Agree)	2 (Agree)	3 (Disagree)	4 (Strongly disagree	
We don't get along well	together.			
We don't talk to each ot	her when we are ang	ry.		
We are generally dissatis	sfied with the family	duties assigned to	us.	
Even though we mean well, we intrude too much into each other's lives.				
There are rules in our family about dangerous situations.				
We confide in each other.				
We cry openly.				
We don't have reasonable transport.				
When we don't like wha	t someone has done,	we tell them.		
We try to think of differe	ent ways to solve pro	blems.		

Appendix II: Parent's Consent Form



University of the Western Cape

Private Bag X 17, Bellville 7535, South Africa Tel: 084 827 26 23 E-mail: suzy.lenders.sl@gmail.com

CONSENT FORM FOR PARENTS

Title of Research Project: The role of family functioning in the decision making styles of adolescents within the Overberg Community

The study has been described to me in a language that I understand and I freely and voluntarily agree to allow my child to participate in the study. My questions about the study have been answered. I understand that my child's identity will not be disclosed and that my child may withdraw from the study without giving a reason at any time and this will not negatively affect my child in any way.

Parent's name	
Parent's signature	UNIVERSITY of the
Witness	WESTERN CAPE
Date	

Should you have any questions regarding this study or wish to report any problems you have experienced related to the study, please contact the study coordinator:

Study Coordinator's Name: Prof. N Roman

University of the Western Cape

Private Bag X17, Belville 7535

Telephone: 021 959 2277/2970

Email: nroman@uwc.ac.za

Appendix III: Assent Form for Participant



University of the Western Cape

Private Bag X 17, Bellville 7535, South Africa Tel: 084 827 26 23 E-mail: suzy.lenders.sl@gmail.com

ASSENT FORM FOR PARTICIPANTS

Title of Research Project: The role of family functioning in the decision making styles of adolescents within the Overberg Community

The study has been described to me in a language that I understand and I freely and voluntarily agree to participate. My questions about the study have been answered. I understand that my identity will not be disclosed and that I may withdraw from the study without giving a reason at any time and this will not negatively affect me in any way.

UNIVERSITY of the

Participant's name	WESTERN CAPE
Participant's signature	
Witness	
Date	

Should you have any questions regarding this study or wish to report any problems you have experienced related to the study, please contact the study coordinator:

Study Coordinator's Name: Prof. N Roman

University of the Western Cape

Private Bag X17, Belville 7535

Telephone: 021 959 2277/2970

Email: nroman@uwc.ac.za

Appendix IV: Information Sheet / Letter



University of the Western Cape

Private Bag X 17, Bellville 7535, South Africa Tel: 084 827 26 23 E-mail: suzy.lenders.sl@gmail.com

INFORMATION SHEET

Project Title: The role of family functioning in the decision making styles of adolescents within the Overberg Community

What is this study about?

This is a research project being conducted by Suzette Rene Lenders of the University of the Western Cape. We are inviting you to participate in this research project because you are a Grade 9 learner at a secondary school in the rural Western Cape. Your participation is voluntary. The purpose of this research project is to examine the role of family functioning in the decision making styles of adolescents, to assist in developing a decision making programme enabling youth to make more effective decisions.

What will I be asked to do if I agree to participate?

You will be asked to complete a questionnaire. This questionnaire will ask you questions about:

- Yourself,
- How you make decisions and determine your preferred decision making style.
- Your family and the overall functioning of your family.

This questionnaire will be completed at school, with permission of your parents, principal and teachers at a time which is not disruptive to your learning. Completion of the questionnaire will be less than 35 minutes.

Would my participation in this study be kept confidential?

Your personal information will be kept confidential as will be totally private. Your name will not be reflected on the questionnaire so you cannot be identified as a participant in this study. The information will be treated with anonymity and confidentiality at all times. The information obtained from the survey will be collated with the information from other completed surveys. Therefore there will be no way to connect you to the survey questionnaire.

What are the risks of this research?

There are no known risks in participating in the study.

What are the benefits of this research?

Information about this topic is limited. This research is not designed to help you personally, but the results may help the investigator learn more about the perceived decision making styles and family functioning of adolescence. Based on this information's limitation in South Africa this study will increase the knowledge for (1) parents, (2) teachers/school, (3) practitioners and (4) the broader society.

Do I have to be in this research and may I stop participating at any time?

Your participation in this research is voluntary as it requires you to choose to take part or not to take part in the study. Should you decide to participate in this research study, you may stop participating at any time and will not be penalised or harmed in any way.

Is any assistance available if I am negatively affected by participating in this study?

Every effort has been taken to protect you from any harm in this study. If however, you may feel affected you can be referred to your nearest community resource for assistance upon request.

What if I have questions?

This research is being conducted by Suzette Rene Lenders registered student of the Social Work Department at the University of the Western Cape. If you have any questions about the research study itself, please contact the study co-ordinator: Prof. N. Roman at: 0219592277/2970 or email: nroman@uwc.ac.za.

Should you have any questions regarding this study and your rights as a research participant or if you wish to report any problems you have experienced related to the study, please contact:

Professor Jose Frantz – Dean of the Faculty of Community and Health Sciences

Tel No: 021 959 2631/2746

Email address: jfrantz@uwc.ac.za

This research has been approved by the University of the Western Cape's Senate Research Committee and Ethics Committee.



Appendix V: Editorial Certificate

5th June 2015

To whom it may concern

Dear Sir/Madam

RE: Editorial Certificate

This letter serves to prove that the thesis listed below was language edited for proper English, grammar, punctuation, spelling as well as overall layout and style by myself, publisher/proprietor of Aquarian Publications, a native English speaking editor.

Thesis title

THE ROLE OF FAMILY FUNCTIONING
IN THE DECISION-MAKING STYLES OF ADOLESCENTS
IN THE OVERBERG AREA

Author

Suzette René Lenders RSITY of the

The research content or the author's intentions were not altered in any way during the editing process, however, the author has the authority to accept or reject my suggestions and changes.

Should you have any questions or concerns about this edited document, I can be contacted at the listed telephone and fax number, e-mail address or website.

Yours truly

EHLondt

Publisher/Proprietor

AQUARIAN PUBLICATIONS

STREET ADDRESS

9 Dartmouth Road Muizenberg 7945

POSTAL ADDRESS

P O Box 00000 Muizenberg 7946

TELEPHONE

021 788 1577

FAX 021 788 1577

MOBILE 076 152 3853

E-MAIL

eddi.aquarian@gmail.com

WEBSITE
www.aquarianpublications.com

PUBLISHER/PROPRIETOR

E H Londt